### UNIVERSITÀ DEGLI STUDI DI PADOVA Department of Information Engineering

### Ph.D. Thesis

# Constrained approximation of spectral densities and spectral estimation

Advisors: Prof. Augusto Ferrante Prof. Michele Pavon

Ph.D. candidate: Federico Ramponi



Dedicated to the Pyrenean Ibex, extinguished on January 6, 2000.

### Notes about this document

Revision: 1.1.3 Compiled: Thu Jan 22 14:03:00 CET 2009

Minimal updates to the present work, such as corrections of typos, may occasionally occur. Please check the latest version of this document at the following address: http://www.dei.unipd.it/~rampo/papers/phdthesis.pdf

# Contents

Sι	Summary					
1	A generalized moment problem					
	1.1	Introduction	3			
		1.1.1 Previous work of the Byrnes, Georgiou and Linquist school	3			
		1.1.2 A generalized moment problem and the search for a solution	4			
		1.1.3 Structure of the work	6			
	1.2	A generalized moment problem	7			
		1.2.1 Existence of solutions	8			
	1.3	Feasibility and the operator $\Gamma$	10			
<b>2</b>	App	proximation in the Kullback-Leibler distance	13			
	2.1	Scalar spectrum approximation	13			
		2.1.1 Optimality conditions	14			
	2.2	An existence theorem	16			
	2.3	The Pavon-Ferrante algorithm	18			
	2.4	Difficulties in extending to the multivariate case	20			
3	App	proximation in the Hellinger distance	23			
	3.1	The Hellinger distance	23			
	3.2	Scalar approximation	25			
		3.2.1 Variational analysis	25			
		3.2.2 The dual problem $\ldots$	27			
	3.3	A naïve multivariable generalization	27			
	3.4	Hellinger distance and spectral factorization	29			
	3.5	$D_H$ -optimal multivariable spectrum approximation	32			
4	A matricial Newton algorithm					
	4.1	The Newton algorithm	41			
	4.2	Hellinger distance minimization with the Newton Algorithm	42			
		4.2.1 Factorization of $Q_{\Lambda}^{-1}(z)$	45			
		4.2.2 Computation of the integrals in $(4.7)$ and $(4.8)$	46			
		4.2.3 Computations in the backtracking step	48			
	4.3	Proof of global convergence	49			

5	<ul> <li>An application to spectrum estimation</li> <li>5.1 A spectral estimation procedure</li></ul>						
		5.2.1	Simulation results: Scalar case	53			
		5.2.2	Simulation results: Multivariate case	55			
A Proofs of some technical lemmas							
Bi	Bibliography						

### Summary

In this work, we consider the problem of finding, among the solutions of a generalized moment problem à la Byrnes/Georgiou/Lindquist, the best approximation to a given spectrum, with respect to different notions of distance. After an in-depth discussion of the moment problem in question, we first review the scalar spectrum approximation in the Kullback-Leibler metric. Then we pose and solve a similar problem with respect to the Hellinger distance. We show that the latter distance admits a nice extension to the case when multivariate spectra come into play, and we solve the resulting, generalized approximation problem. Finally, we present in detail a matricial version of the Newton algorithm designed to solve the corresponding dual problem, that eventually leads to an useful solution, and provide an application to multivariate spectrum estimation, testing it against well-known identification methods.

This work contains the result of three years of collaboration between my advisors Augusto Ferrante and Michele Pavon, and me. In essence, it is an assembly of four papers we have written together, namely [22], [21], [20], and [51], with minor corrections and improvements. Without their teaching, support, and continuous encouragement, I would have published nothing, and this thesis would consist of a few blank pages. For the time spent together, to them goes my gratitude.

Also, we wish to thank Prof. Anders Lindquist and Prof. Tryphon Georgiou for an enduring, enlightening correspondence that helped us develop our results, and for their willingness to suggest and discuss these subjects during the time they were in Padova.

Padova, January 20, 2009

Federico Ramponi

### Chapter 1

### A generalized moment problem

### 1.1 Introduction

### 1.1.1 Previous work of the Byrnes, Georgiou and Linquist school

ARMA identification methods usually lead to nonconvex optimization problems for which global convergence is not guaranteed, cf. e.g. [43, 55, 56, 17]. Although these algorithms are simple and perform effectively, as observed in [56, p.103], [42, Section 1], no theoretically satisfactory approach to ARMA parameter estimation appears to be available. Alternative, convex optimization approaches have been recently proposed by Byrnes, Georgiou, Lindquist and co-workers [6, 36] in the frame of a broad research effort on analytic interpolation with degree contraint, see [3, 5, 8, 9, 10, 11, 12, 7, 18, 28, 24, 25, 26, 27, 29, 30, 32, 33, 34, 35] and references therein. In particular, [7] describes a new setting for spectral estimation. The so-called *THREE* algorithm introduced there appears to allow for higher resolution in prescribed frequency bands and to be particularly suitable in case of short observation records. It effectively detects spectral lines and steep variations (see [47] for a recent biomedical application). An outline of this method is as follows. A given realization of a stochastic process (a finite collection of data  $y_1...y_N$ ) is fed to a suitably structured bank of filters, and the steady-state covariance matrix of the resulting output is estimated by statistical methods. Only zeroth-order covariance lags of the output of the filters need to be estimated, ensuring statistical robustness of the method. Finding now an input process whose *rational* spectrum is compatible with the estimated covariance poses naturally a Nevanlinna-Pick interpolation problem with bounded degree. The solution of this interpolation problem is considered as a mean of estimating the spectrum. A particular case described in the paper is the maximum differential entropy spectrum estimate, which amounts to the so-called central solution in the Nevanlinna-Pick theory. More generally, the scheme allows for a non constant *a priori* estimate  $\Psi$  of the spectrum. The Byrnes-Georgiou-Lindquist school has shown how this and other important problems of control theory may be advantageously cast in the frame

of convex optimization. These problems admit a finite dimensional dual (multipliers are matrices!) that can be shown to be solvable. The latter result, due to Byrnes and Lindquist [12] (see also [20]) is, however, nontrivial since the optimization occurs on an open, unbounded set of Hermitian matrices. The numerical solution of the dual problem is also challenging [7, 18, 46], since the gradient of the dual functional tends to infinity at the boundary of the feasible set. Finally, reparametrization of the problem may lead to loss of global concavity, see the discussion in [34, Section VII].

This work adds to this effort in that we consider estimation of a multivariate spectral density in the spirit of THREE [7], but employing a different metric for the optimization part, namely the *Hellinger distance* as in [21]. In papers [10, 11], Byrnes, Gusev and Lindquist chose the Kullback-Leibler divergence as a frequency weighted entropy measure, thus introducing a broad generalization of Burg's maximum entropy method. More recently, this motivation was supported by the well-known connection with prediction error methods, see e.g. [57, 42]. In the multivariable case, a Kullback-Leibler pseudodistance may also be readily defined [30] inspired by the *von Neumann's relative entropy* [60, 59] of statistical quantum mechanics. The resulting spectrum approximation problem, however, leads to computable solutions of bounded McMillan degree only in the case when the prior spectral density is the identity matrix [30, 21] (maximum entropy solution). On the contrary, with a suitable extension of the scalar Hellinger distance introduced in [21], the Hellinger approximation generalizes nicely to the multivariable case for any prior estimate  $\Psi$  of the spectrum.

# 1.1.2 A generalized moment problem and the search for a solution

The method we propose to estimate a multivariate spectrum relies on the solution of a *constrained spectrum approximation* problem. Let us explain in simple words what this means.

Consider the following situation: a discrete-time, wide-sense stationary, *m*-dimensional random process y is fed to an asymptotically stable system with transfer function  $G(z) = (zI - A)^{-1}B$ . The asymptotic state covariance matrix of the system will be:

$$\Sigma = \int_{-\pi}^{\pi} G(e^{j\vartheta}) \Phi(e^{j\vartheta}) G^{\top}(e^{-j\vartheta}) \frac{\mathrm{d}\vartheta}{2\pi}$$
(1.1)

where  $\Phi$  is the spectrum of y. We consider the following moment problem:

#### Given G and $\Sigma$ , find a spectral density $\Phi$ for which (1.1) holds.

Moment problems are generally considered as "inverse" problems (where the term "inverse" stems from the observation that the simple computation of  $\Sigma$ , given G and  $\Phi$ , should be regarded as "direct"). Recall that a problem is said to be *well posed*, in the sense of Hadamard, if the following properties hold:

• a solution to the problem exists;

- such solution is unique;
- the solution depends continuously on the data.

Inverse problems such as the one we are considering (where "data" means the pair  $G, \Sigma$ ) are typically *not* well posed. In particular, there may be no solution, and when a solution exists, usually there are infinitely many.

Thus, given  $(G, \Sigma)$ , the question arises as whether there are solutions  $\Phi$  to (1.1), and if this is the case, to find and eventually parameterize the family of such solutions. Being geared toward system-theoretic applications, we further ask whether *rational* solutions exist and, moreover, if the complexity of such solutions (their MacMillan degree) can be bounded *a priori*. A full answer to these questions has been given by Georgiou, Lindquist and their collaborators.

Let us change our point of view from a theoretical to a practical perspective. Suppose that we can choose  $G(z) = (zI - A)^{-1}B$ , with the only constraints of A being a stability matrix and (A, B) being reachable. Here G(z) models a bank of filters. Now a realization of the system with transfer function G(z) is "physically there", and we feed it with a finite chunk of the process y, say  $\{y_1, ..., y_N\}$ . Observing the states of the system, say  $\{x_1, ..., x_N\}$ , we then compute a Hermitian and positive definite estimate  $\hat{\Sigma}$  of the asymptotic state covariance. Given G, our question is now how to exploit the estimated  $\hat{\Sigma}$  to make inferences about the true spectrum  $\Phi$  of y. That is, we want to provide an estimate of  $\Phi$ , taking into account that the state variance we have obtained is  $\hat{\Sigma}$ . The issues are manifold:

- 1. The set of the  $\Sigma$ 's for which a solution exists is, in general, a thin set. Therefore, if we set  $\Sigma = \hat{\Sigma}$ , (1.1) will be usually unfeasible. Thus, we need to find a second estimate  $\bar{\Sigma}$ , near to the first, that belongs to this set.
- 2. Having done so, the solutions  $\Phi$  to problem (1.1) with  $\Sigma = \overline{\Sigma}$  will however be infinitely many. Thus, in order to make inferences about  $\Phi$ , we will have to choose *one* solution  $\hat{\Phi}$ , satisfying some optimality criterion, and elect it as an estimate of  $\Phi$ .
- 3. Moreover, for practical purposes we require this particular solution to be rational and with bounded degree.
- 4. Finally, we may ask if tuning G(z) we can somehow obtain better inferences about  $\Phi$ .

This work is mainly devoted to the second issue, and precisely to a suitable choice of the optimality criterion.

One more or less obvious such choice is the maximum entropy principle. Adding milestones to a path that was pioneered by Burg, in [27] Georgiou has provided an explicit expression for the spectrum  $\hat{\Phi}$  that exhibits maximum entropy rate among the solutions of (1.1), when they exist.

But, following [34], let us add another ingredient to the problem. Suppose that some information about  $\Phi$  becomes available under the form of a *prior* spectrum  $\Psi$ . That is, suppose we are said that the spectrum  $\Psi$  is "near" to  $\Phi$ , for some notion of distance. Now given G,  $\Sigma$ , and  $\Psi$ , we search the spectrum  $\hat{\Phi}$ , which is closest to  $\Psi$ , among the solutions of (1.1). In the following we choose this as the optimality criterion, and we call the search for such a solution a spectrum approximation problem, subject to the constraint (1.1).

The paper [34] poses and solves a variational problem in this very spirit. It deals with the case when y is a scalar process, and employs the minimization of the Kullback-Leibler pseudo-distance from  $\Psi$  to  $\Phi$  as the optimality criterion. This approach has relevant advantages, namely that it leads to a simple, rational expression for the optimal solution, and with bounded degree. Moreover, it admits Georgiou's maximum entropy solution as a particular case ( $\Psi \equiv 1$ , when y is scalar). But it has a major drawback: For any notion of multivariable extension of the Kullback-Leibler pseudodistance that can be regarded as "natural", the analysis that is proposed there seems impossible to be carried through. That is, this approach cannot be employed in the multivariate case.

The main contribution of this work, which in essence is a rewriting of papers [21] and [51], is a multivariable extension of another metric, namely the Hellinger distance, in which the variational analysis can be carried through, and is relatively simple. The spectrum  $\hat{\Phi}$  which is closest to  $\Psi$  with respect to this distance is again rational, and with bounded degree.

### 1.1.3 Structure of the work

The rest of Chapter 1 contains a more precise statement of the moment problem in question, and some properties of the vector space of the matrices  $\Sigma$  for which it admits solutions.

Chapter 2 presents a review of spectrum approximation in the Kullback-Leibler pseudo-distance, and an explanation of why is difficult to extend it to the multivariable case.

Chapter 3 introduces the Hellinger distance, and presents the variational analysis that leads to the minimizing spectrum both in the scalar and in the multivariable case. In either setting, however, the optimal spectrum depends on the Lagrange parameter  $\Lambda$ . To meet the constraint (1.1) and come to a true solution, one must pursue the solution of the *dual problem*, which is now finite-dimensional.

Chapter 4 contains a detailed description of a matricial version of the Newton algorithm suited to solve the dual problem.

Finally, Chapter 5 presents an application to multivariable spectrum estimation, tests it through simulation, and deals in some measure with the first and fourth issues of the previous section, namely the choice of a suitable estimate of  $\Sigma$  and the tuning of G(z).

### 1.2 A generalized moment problem

We consider the following basic set-up patterned after [27, 34, 30]. Let  $\mathcal{S}^{m\times m}_+(\mathbb{T})$  be the family of bounded, coercive,  $\mathbb{C}^{m\times m}$ -valued spectral density functions on the unit circle. Thus, a measurable, bounded matrix valued function  $\Phi$  belongs to  $\mathcal{S}^{m\times m}_+(\mathbb{T})$  if it satisfies the following properties:

- the values of  $\Phi$  are  $m \times m$ , Hermitian, non negative definite matrices;
- there exists a positive constant  $c_{\Phi}$  such that  $\Phi(e^{i\vartheta}) c_{\Phi}I$  is positive definite a.e. on  $\mathbb{T}$ .

Notice that  $\Phi \in \mathcal{S}^{m \times m}_{+}(\mathbb{T})$  if and only if  $\Phi^{-1} \in \mathcal{S}^{m \times m}_{+}(\mathbb{T})$ . Let  $\Psi \in \mathcal{S}^{m \times m}_{+}(\mathbb{T})$  represent an *a priori* estimate of the spectrum of an underlying zero-mean, wide-sense stationary *m*-dimensional stochastic process  $\{y(n), n \in \mathbb{T}\}$ . We consider a rational transfer function

$$G(z) = (zI - A)^{-1}B, \qquad A \in \mathbb{C}^{n \times n}, B \in \mathbb{C}^{n \times m}, \tag{1.2}$$

where A is a stability matrix, B is full column rank, and (A, B) is a reachable pair. Suppose moreover that we *know* the asymptotic state covariance  $\Sigma > 0$  of the system with transfer function G and input the unknown process y. In other words, we suppose we know the covariance of the n- dimensional stationary process  $\{x_k; k \in \mathbb{T}\}$  satisfying

$$x_{k+1} = Ax_k + By_k, \quad k \in \mathbb{T}.$$
(1.3)

In general,  $\Psi$  is not consistent with  $\Sigma$ , and it is necessary to find  $\Phi$  in  $\mathcal{S}^{m \times m}_{+}(\mathbb{T})$  that is closest to  $\Psi$  in a suitable sense among spectra consistent with  $\Sigma$ , namely satisfying

$$\int G\Phi G^* = \Sigma, \tag{1.4}$$

where star denotes transposition plus conjugation. Here, and throughout the work, integration takes place on  $[-\pi, \pi]$  with respect to the normalized Lebesgue measure  $d\vartheta/2\pi$ , i.e.:

$$\int f := \int_{-\pi}^{\pi} f(\mathrm{e}^{\mathrm{j}\vartheta}) \, \frac{\mathrm{d}\vartheta}{2\pi}$$

The question of existence of  $\Phi \in \mathcal{S}^{m \times m}_{+}(\mathbb{T})$  satisfying (1.4) and, when existence is granted, the parametrization of all solutions to (1.4), may be viewed as a generalized moment problem.

Here, "generalized" stands for the fact that the *classical* moment problem, where we search  $\Phi \in \mathcal{S}^{1\times 1}_+(\mathbb{T})$  under constraints on its moments  $\int \Phi$ ,  $\int e^{j\vartheta} \Phi$ ,  $\int e^{2j\vartheta} \Phi$ , ..., is a particular case of this. More precisely, in the case m = 1, take G(z) with k-th component  $G_k(z) = z^{k-n-1}$ . Take moreover

$$A = \begin{bmatrix} 0 & 1 & 0 & \dots & 0 \\ 0 & 0 & 1 & \dots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \dots & 1 \\ 0 & 0 & 0 & \dots & 0 \end{bmatrix}, \quad B = \begin{bmatrix} 0 \\ 0 \\ \vdots \\ 0 \\ 1 \end{bmatrix},$$

$$\Sigma = \begin{bmatrix} c_0 & c_1 & c_2 & \dots & c_{n-1} \\ \bar{c}_1 & c_0 & c_1 & \ddots & c_{n-2} \\ \bar{c}_2 & \ddots & \ddots & \ddots & \ddots \\ \vdots & \ddots & \ddots & \ddots & \ddots \\ \bar{c}_{n-1} & \bar{c}_{n-2} & \ddots & \ddots & c_0 \end{bmatrix},$$
(1.5)

where  $c_k := E\{y(n)\overline{y}(n+k)\}$ . This is the covariance extension problem, where the information available on the process y is the finite sequence of covariance lags  $c_0, c_1, \ldots, c_{n-1}$ . It is known that the set of densities consistent with the data is nonempty if  $\Sigma \ge 0$  and contains infinitely many elements if  $\Sigma > 0$  (see for instance [37]).

As another example, again in the case m = 1, take  $G_k(z) = \frac{1}{z - p_k}$ , or

$$A = \begin{bmatrix} p_1 & 0 & 0 & \dots & 0 \\ 0 & p_2 & 0 & \dots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \dots & 0 \\ 0 & 0 & 0 & \dots & p_n \end{bmatrix}, \quad B = \begin{bmatrix} 1 \\ 1 \\ \vdots \\ 1 \\ 1 \end{bmatrix}$$
(1.6)

where the  $p_i$ 's lie inside the unit circle, and let  $\Sigma$  be the Pick matrix with elements

$$\Sigma_{i,j} = \frac{w_i + \bar{w}_j}{1 - p_i \bar{p}_j}$$

where

$$w_{k} = \frac{1}{4\pi} \int_{-\pi}^{\pi} \frac{e^{-i\theta} + p_{k}}{e^{-i\theta} - p_{k}} \Phi(e^{i\theta}) d\theta, \quad k = 1, 2, \dots, n.$$

In this case the problem is a Nevanlinna-Pick interpolation problem. Solutions exist if and only if  $\Sigma \geq 0$ , and the solution is unique if and only if  $\Sigma$  is singular.

#### **1.2.1** Existence of solutions

Existence of  $\Phi \in \mathcal{S}^{m \times m}_+(\mathbb{T})$  satisfying constraint (1.4) is a nontrivial issue. It was shown in [28] that such family is nonempty if and only if there exists  $H \in \mathbb{C}^{m \times n}$  such that

$$\Sigma - A\Sigma A^* = BH + H^* B^*, \tag{1.7}$$

or, equivalently, the following rank condition holds:

$$\operatorname{rank}\left(\begin{array}{cc} \Sigma - A\Sigma A^* & B\\ B^* & 0 \end{array}\right) = 2m.$$
(1.8)

**Remark 1.2.1** Since we develop our theory under the assumption that  $\Sigma > 0$ , without loss of generality we can always assume  $\Sigma = I$ . Indeed, if  $\Sigma \neq I$ , it suffices to replace G with  $G' := \Sigma^{-1/2}G$  and (A, B) with  $(A' = \Sigma^{-1/2}A\Sigma^{1/2}, B' = \Sigma^{-1/2}B)$ . Thus, throughout the rest of the work, (1.4) will read

$$\int G\Phi G^* = I. \tag{1.9}$$

We wish to give an alternative formulation of the existence result (1.7). Let  $\Pi_B = B(B^*B)^{-1}B^*$  denote the orthogonal projection onto Range B.

**Proposition 1.2.2** A necessary and sufficient condition for the existence of spectra in  $\mathcal{S}^{m \times m}_{+}(\mathbb{T})$  satisfying (1.9) is that the following relation holds

$$(I - \Pi_B) (I - AA^*) (I - \Pi_B) = 0.$$
(1.10)

When (1.10) is satisfied, there exists  $\Phi \in \mathcal{S}^{m \times m}_+(\mathbb{T})$  satisfying (1.9) of McMillan degree less than or equal to 2n.

*Proof. Necessity:* Suppose there exists y *m*-dimensional, wide-sense stationary with spectral density  $\Phi \in \mathcal{S}^{m \times m}_{+}(\mathbb{T})$  satisfying (1.9). Let x be defined by (1.3). Taking covariances on both sides of (1.3), we get

$$I = AA^* + AE\{x_k y_k^*\}B^* + BE\{y_k x_k^*\}A^* + BE\{y_k y_k^*\}B^*.$$

Now taking  $AA^*$  to the left-hand side of the equation, and pre- and post-multiplying each side by  $(I - \Pi_B)$ , we obtain (1.10).

Sufficiency: We adapt the argument in [27, p.1814]. For a given purely non deterministic *m*-dimensional process y with spectrum  $\Phi$ , define the process w as the output of the linear stable system

$$\begin{aligned} x_{k+1} &= Ax_k + By_k, \\ w_k &= (B^*B)^{-1}B^*x_{k+1} \end{aligned}$$
(1.11)

Inverting the system (1.11), we get

$$\begin{aligned} x_{k+1} &= (I - \Pi_B)Ax_k + Bw_k, \\ y_k &= -(B^*B)^{-1}B^*Ax_k + w_k. \end{aligned}$$
(1.12)

Write (1.10) as a Lyapunov identity

$$I = (I - \Pi_B) A A^* (I - \Pi_B) + \Pi_B.$$
(1.13)

Since (A, B) is controllable, so is the pair  $((I - \Pi_B)A, B(B^*B)^{-1/2})$ . It now follows from (1.13) that  $(I - \Pi_B)A$  has all eigenvalues in the open unit disc  $\mathbb{D}$ . Thus, system (1.12) is stable,  $(B^*B)^{-1}B^*G(z)$  is minimum phase and the processes y and w are causally equivalent. It follows that if we choose w to be a white noise sequence with intensity  $E\{w_k w_k^*\} = (B^*B)^{-1}$  and y to be defined by (1.12) then:

- 1. (1.12) is the innovation representation of y;
- 2. the state covariance of the steady-state Kalman filter (1.12) satisfies the Lyapunov equation (1.13) and is therefore the identity;
- 3. the spectral density of y is given by

$$\Phi_y = W(z)(B^*B)^{-1}W(z)^*, \qquad (1.14)$$

where

$$W(z) = I - (B^*B)^{-1}B^*A (zI - (I - \Pi_B)A)^{-1}B,$$

is the transfer function of (1.12).

We conclude that if we feed G in (1.11) with such a process y, the filter state x will have the required covariance, namely the identity matrix, and (1.9) will be satisfied. Moreover,  $\Phi_y$  is rational of McMillan degree at most 2n and it belongs to  $\mathcal{S}^{m \times m}_+(\mathbb{T})$ since its values and the values of  $\Phi_y^{-1}$  on  $\mathbb{T}$  are positive definite matrices.

The geometric condition (1.10) seems more amenable to generalization than (1.8). The spectrum (1.14) has been shown in [27, Section III] to be the maximum entropy spectrum among those satisfying (1.9). This is there accomplished in a clever way, by relating the constrained maximum entropy problem to a special one-step-ahead prediction problem.

### **1.3** Feasibility and the operator $\Gamma$

In this section, we discuss in depth the feasibility of (1.4). We adopt the following notation for the vector space of all the  $n \times n$  Hermitian matrices:

$$\mathcal{H}(n) := \{ M \in \mathbb{C}^{n \times n} : M = M^* \}.$$

$$(1.15)$$

Let  $\mathcal{C}(\mathbb{T}; \mathcal{H}(m))$  be the space of  $\mathcal{H}(m)$ -valued continuous functions defined on the unit circle, and let the operator  $\Gamma : \mathcal{C}(\mathbb{T}; \mathcal{H}(m)) \to \mathcal{H}(n)$  be defined as follows:

$$\Gamma(\Phi) := \int G \Phi G^* \tag{1.16}$$

We are interested in the *range* of the operator  $\Gamma$  which, having to deal with Hermitian matrices, we consider as a vector space over the reals.

#### **Proposition 1.3.1** The following facts hold:

- 1. Let  $\Sigma = \Sigma^* > 0$ . The following are equivalent:
  - There exists  $H \in \mathbb{C}^{m \times n}$  such that identity (1.7) holds.
  - There exists  $\Phi \in \mathcal{S}^{m \times m}_+(\mathbb{T})$  such that  $\int G \Phi G^* = \Sigma$ .
  - There exists  $\Phi \in \mathcal{C}(\mathbb{T}; \mathcal{H}(m)), \Phi > 0$  such that  $\Gamma(\Phi) = \Sigma$ .
- 2. Let  $\Sigma = \Sigma^*$  (not necessarily positive definite). There exists  $H \in \mathbb{C}^{m \times n}$  such that identity (1.7) holds if and only if  $\Sigma \in \text{Range }\Gamma$ .
- 3.  $X \in \operatorname{Range} \Gamma^{\perp}$  if and only if  $G^*(e^{j\vartheta})XG(e^{j\vartheta}) = 0 \ \forall \vartheta \in [0, 2\pi].$

Proof. As stated above, it was proved in [28] that there exists  $H \in \mathbb{C}^{m \times n}$  such that identity (1.7) holds with Hermitian and positive definite  $\Sigma$  if and only if  $\Sigma = \int G \Phi G^*$ for some  $\Phi \in \mathcal{S}^{m \times m}_+(\mathbb{T}), \Phi > 0$ . The proof of Fact 1 is straightforward once we note that the "if" part of the proof of Proposition 1.2.2 is *constructive*, and exhibits a *continuous* spectrum. Hence, the fact that there exists a spectrum  $\Phi$  such that  $\Sigma = \int G \Phi G^*$  is equivalent to there exists a *continuous* spectrum such that the same holds.

As for the second assertion, let  $\Sigma \in \text{Range }\Gamma$ . Then there exists  $\Phi \in \mathcal{C}(\mathbb{T}; \mathcal{H}(m))$  such that

$$\Sigma = \int G\Phi G^* = \int G(\Phi_+ - \Phi_-)G^*$$
$$= \int G\Phi_+ G^* - \int G\Phi_- G^* = \Sigma_+ - \Sigma_-$$

where  $\Phi_+$  and  $\Phi_-$  are two spectra such that  $\Phi_+ - \Phi_- = \Phi$  (they can be chosen to be bounded away from zero) and where  $\Sigma_+$  and  $\Sigma_-$  are symmetric positive definite. Hence  $\Sigma$  is a difference of *positive* matrices for which (1.7) holds. This establishes (1.7) for  $\Sigma$  itself. Vice versa, suppose that (1.7) holds for an Hermitian  $\Sigma$ . Let  $\Sigma_{\alpha}$  be the unique solution of the following Lyapunov equation:

$$\Sigma_{\alpha} - A\Sigma_{\alpha}A^* = B\left(\alpha B^*\right) + \left(\alpha B^*\right)^* B^* = 2\alpha BB^*$$

where  $\alpha \in \mathbb{R}$ . Then  $\Sigma_{\alpha}$  depends linearly upon  $\alpha$ , i.e.  $\Sigma_{\alpha} = \alpha \Sigma_1$ , where  $\Sigma_1 > 0$  since (A, B) is reachable. Thus, there exists an  $\alpha$  such that  $\Sigma_{\alpha} > 0$  and  $\Sigma_{\alpha} > \Sigma$ . Let  $\Sigma_{-} = \Sigma_{\alpha} - \Sigma$ . Then  $\Sigma_{-} > 0$ , and since (1.7) holds for  $\Sigma$  and  $\Sigma_{\alpha}$ , it also holds for  $\Sigma_{-}$ . Then assertion 1 implies that there exist  $\Phi_{\alpha} > 0$  and  $\Phi_2 > 0$  in  $\mathcal{C}(\mathbb{T}; \mathcal{H}(m))$  such that  $\Sigma_{\alpha} = \int G \Phi_{\alpha} G^*$  and  $\Sigma_{-} = \int G \Phi_2 G^*$ , hence  $\Sigma = \int G (\Phi_{\alpha} - \Phi_2) G^*$  and assertion 2 follows.

The third assertion is a simple geometrical fact: If  $X \in \text{Range }\Gamma^{\perp}$ , then for any  $\Phi \in \mathcal{C}(\mathbb{T}; \mathcal{H}(m))$ 

$$0 = \left\langle X, \int G\Phi G^* \right\rangle = \operatorname{tr} X \int G\Phi G^* = \operatorname{tr} \int (G^*XG)\Phi$$

and the conclusion follows.

**Remark 1.3.2** The underlying statement in Proposition 1.3.1, Facts 1 and 2, is that if we defined  $\Gamma$  over the vector space of finite linear combinations of functions in  $S^{m\times m}_+(\mathbb{T})$ , its range would remain the same. Let us restate this in another way. Let  $\overline{\Gamma}$  be an extension of  $\Gamma$  to span  $S^{m\times m}_+$ , defined exactly as in (1.16). Given a spectrum  $\Phi \in \operatorname{span} S^{m\times m}_+$  (recall that this implies that  $\Phi$  is bounded), there exists a uniformly bounded sequence of continuous functions  $\{\Phi^n_c\} \subset C(\mathbb{T}; \mathcal{H}(m))$  that converges to  $\Phi$  in the  $L^1$  topology. Then there exists a subsequence  $\{\Phi^{n_i}_c\}$  that converges to  $\Phi$  pointwise, and by dominated convergence it holds  $\lim_{i\to\infty} ||\overline{\Gamma}(\Phi^{n_i}_c) - \overline{\Gamma}(\Phi)|| = 0$ . Thus, any  $\Sigma \in \operatorname{Range} \overline{\Gamma}$  is a limit point of  $\overline{\Gamma}(\mathcal{C}(\mathbb{T}; \mathcal{H}(m))) \equiv \operatorname{Range} \Gamma$ . But Range  $\Gamma$ , being finite dimensional, hence closed, already contains all its limit points.

**Remark 1.3.3** Proposition 1.3.1 shows that Range  $\Gamma$  is the set of all the Hermitian matrices  $\Sigma$  for which there exists H such that (1.7) holds. This fact will be useful in numerical computations. Indeed, Range  $\Gamma$  is obviously finite-dimensional, and if  $\{H_1, ..., H_N\}$  is a base of  $\mathbb{C}^{m \times n}$ , then the corresponding solutions  $\{\Sigma_1, ..., \Sigma_N\}$  of (1.7), considered as a discrete-time Lyapunov equation in the unknown  $\Sigma$ , generate Range  $\Gamma$ . Note that  $\{\Sigma_1, ..., \Sigma_N\}$  are not necessarily linearly independent.

### Chapter 2

# Approximation in the Kullback-Leibler distance

### 2.1 Scalar spectrum approximation

In [34], the Kullback-Leibler measure of distance for spectra in  $\mathcal{S}_+(\mathbb{T}) := \mathcal{S}_+^{1\times 1}(\mathbb{T})$  was introduced:

$$\mathbb{D}(\Psi \| \Phi) = \int \Psi \log \left(\frac{\Psi}{\Phi}\right)$$

As is well known, this pseudo-distance originates in hypothesis testing, where it represents the mean information for observation for discrimination of an underlying probability density from another [41]. It also plays a central role in information theory, identification, stochastic processes, etc., see e.g. [39, 16, 15, 23, 2, 13, 59, 50] and references therein. It is also known in these fields as *divergence*, *relative entropy*, *information distance*. etc. If

$$\int \Phi = \int \Psi,$$

we have  $\mathbb{D}(\Psi \| \Phi) \geq 0$ . The choice of  $\mathbb{D}(\Psi \| \Phi)$  as a distance measure, even for spectra that have different zeroth moment, is discussed in [34, Section III]. It is observed there that the constraint (1.4) often fixes the zeroth Fourier coefficient of feasible spectra (this happens for sure when A is singular). In that case, rescaling  $\Psi$ , we are guaranteed that the index is nonnegative and equal to zero if and only if the two spectra are equal. T. Georgiou has kindly informed us that even when A is nonsingular, under a rather mild assumption, it is possible to modify the index so that all  $\Phi$  satisfying the constraint have the same zeroth moment. In any case, the method entails a rescaling of the *a priori* density  $\Psi$ , so that the optimization problem amounts to approximating the "shape" of the *a priori* spectrum. This is of course sensible to pursue in several engineering applications such as speech processing.

We mention that, in the same spirit, Georgiou has very recently investigated other distances for power spectra, [32, 33]. Motivated by classical prediction theory, where

the optimal one step ahead predictor does not depend on the  $L^1$  norm of the spectrum, he seeks natural distances between *rays* of spectral densities. Considering the degradation of performance when an optimal predictor for one stochastic process is employed to predict a different stochastic process, he is naturally led to introduce a certain metric on rays.

Note that minimizing  $\Phi \to \mathbb{D}(\Psi \| \Phi)$  rather than  $\Phi \to \mathbb{D}(\Phi \| \Psi)$  is unusual with respect to the statistics-probability-information theory world. Besides leading to a more tractable form of the optimal solution, however, it also includes as special case  $(\Psi \equiv 1)$  maximization of entropy [27]. In [34], the following problem is considered:

**Problem 2.1.1** Given  $\Psi \in S_+(\mathbb{T})$ , find  $\hat{\Phi}$  that solves

minimize 
$$\mathbb{D}(\Psi \| \Phi)$$
  
over  $\left\{ \Phi \in \mathcal{S}_{+}(\mathbb{T}) \mid \int G \Phi G^{*} = I \right\}.$  (2.1)

**Remark 2.1.2** In the context of the covariance extension problem (1.5), the minimizers in Problem 2.1.1, when  $\Psi$  ranges over positive trigonometric polynomials of degree n, are precisely the coercive spectra consistent with the first n covariance lags and of degree at most 2n, [10, 11, 24, 25]. This illustrates the role of the "a priori parameter"  $\Psi$  in obtaining a description of all solution to the moment problem of prescribed complexity.

#### 2.1.1 Optimality conditions

Consider Problem 2.1.1. With minor notational differences, the variational analysis in [34] is outlined as follows (see also [49]). Let

$$\mathcal{L}^{KL} := \{ \Lambda \in \mathcal{H}(n) \mid G^* \Lambda G > 0, \forall e^{i\vartheta} \in \mathbb{T} \}.$$
(2.2)

For  $\Lambda \in \mathcal{L}^{KL}$ , consider the Lagrangian function

$$L(\Phi, \Lambda) = \mathbb{D}(\Psi \| \Phi) + \left\langle \Lambda, \int G \Phi G^* - I \right\rangle$$
  
=  $\mathbb{D}(\Psi \| \Phi) + \operatorname{tr} \left( \Lambda \left( \int G \Phi G^* - I \right) \right)$   
=  $\mathbb{D}(\Psi \| \Phi) + \int G^* \Lambda G \Phi - \operatorname{tr}(\Lambda)$  (2.3)

Consider the unconstrained minimization of the strictly convex functional  $L(\Phi, \Lambda)$ :

minimize{
$$L(\Phi, \Lambda) \mid \Phi \in \mathcal{S}_{+}(\mathbb{T})$$
} (2.4)

This is a convex optimization problem. The variational analysis yields the following result.

**Theorem 2.1.3** The unique solution  $\hat{\Phi}$  to problem (2.4) is given by

$$\hat{\Phi} = \frac{\Psi}{G^* \Lambda G}.$$
(2.5)

Moreover, suppose  $\hat{\Lambda} \in \mathcal{L}^{KL}$  is such that

$$\int G \; \frac{\Psi}{G^* \hat{\Lambda} G} \; G^* = I. \tag{2.6}$$

Then  $\hat{\Phi}$  given by

$$\hat{\Phi} = \frac{\Psi}{G^* \hat{\Lambda} G} \tag{2.7}$$

is the unique solution of the approximation Problem (2.1.1).

Thus, the original Problem 2.1.1 is now reduced to finding  $\hat{\Lambda} \in \mathcal{L}^{KL}$  satisfying (2.6). This is accomplished via duality theory. Consider the dual functional

 $\Lambda \mapsto \inf \{ L(\Phi, \Lambda) \mid \Phi \in \mathcal{S}_+(\mathbb{T}) \}.$ 

For  $\Lambda \in \mathcal{L}^{KL}$ , the dual functional takes the form

$$\Lambda \mapsto L\left(\frac{\Psi}{G^*\Lambda G},\Lambda\right) = \int \Psi \log G^*\Lambda G - \operatorname{tr}(\Lambda) + \int \Psi.$$
(2.8)

Consider now the maximization of the dual functional (2.8) over  $\mathcal{L}^{KL}$ . Let, as in [34],

$$J_{\Psi}(\Lambda) := -\int \Psi \log G^* \Lambda G + \operatorname{tr}(\Lambda)$$
(2.9)

The dual problem is then equivalent to

minimize 
$$\{J_{\Psi}(\Lambda) \mid \Lambda \in \mathcal{L}^{KL}\}.$$
 (2.10)

The dual problem is also a convex optimization problem. In [34],  $\Lambda$  is further restricted to belong to the range of the operator  $\Gamma$  defined in section 1.3 (equation (1.16)). The problem then becomes

minimize 
$$\{J_{\Psi}(\Lambda) \mid \Lambda \in \mathcal{L}_{\Gamma}^{KL}\}$$
 (2.11)

where

$$\mathcal{L}_{\Gamma}^{KL} := \mathcal{L}^{KL} \cap \operatorname{Range} \Gamma.$$
(2.12)

The reason for this is that, in the Lagrangian (2.3), the term  $\int G\Phi G^*$  between brackets belongs to Range  $\Gamma$  by construction, while *I* belongs to Range  $\Gamma$  by the feasibility assumption. Then it is natural, though not strictly necessary, to restrict also  $\Lambda$  to Range  $\Gamma$ , thus excluding any component belonging to Range  $\Gamma^{\perp}$ . It should be clear from fact 3 in Proposition 1.3.1 that any such component would not play any role in the Lagrangian.

The functional  $J_{\Psi}$  is shown in [34] to be strictly convex on the restricted domain  $\mathcal{L}_{\Gamma}^{KL}$ . It is also shown in [12] that  $J_{\Psi}$  has a unique minimum point in  $\mathcal{L}_{\Gamma}^{KL}$ . This result implies that, under the feasibility assumption, there exists a (unique)  $\hat{\Lambda}$  in  $\mathcal{L}_{\Gamma}^{KL}$  satisfying (2.6). Such a  $\hat{\Lambda}$  then provides the optimal solution of the primal problem (2.1.1) via (2.7).

### 2.2 An existence theorem

In [34] the existence of a minimum of  $J_{\Psi}$  on  $\mathcal{L}_{\Gamma}^{KL}$  is established resorting to a profound homeomorphism result for continuous maps between open, connected sets of the same dimension. In this section we want to show a different proof, based on less abstract results.

Let us consider the closure of  $\mathcal{L}^{KL}$ , given by

$$\overline{\mathcal{L}^{KL}} = \{ \Lambda = \Lambda^* \in \mathbb{C}^{n \times n} : \ G^* \Lambda G \ge 0, \forall e^{i\vartheta} \in \mathbb{T} \},\$$

and the closure of  $\mathcal{L}_{\Gamma}^{KL}$ , given by

$$\overline{\mathcal{L}_{\Gamma}^{KL}} = \overline{\mathcal{L}^{KL}} \cap \operatorname{Range} \Gamma$$
(2.13)

On the convex set  $\overline{\mathcal{L}_{\Gamma}^{KL}}$ , we define the sequence of functions

$$J_{\Psi}^{n}(\Lambda) := \operatorname{tr}(\Lambda) - \int \Psi \log\left(G^*\Lambda G + \frac{1}{n}\right).$$
(2.14)

**Lemma 2.2.1** The pointwise limit  $J_{\Psi}^{\infty}(\Lambda) = \lim_{n \to \infty} J_{\Psi}^{n}(\Lambda)$  exists and defines a lower semicontinuous, convex function on  $\overline{\mathcal{L}_{\Gamma}^{KL}}$  with values in the extended reals.

<u>Proof.</u> For each n,  $J_{\Psi}^{n}$  is a continuous, convex function on the closed convex set  $\overline{\mathcal{L}_{\Gamma}^{KL}}$ . Hence epi $(J_{\Psi}^{n})$ , the epigraph of  $J_{\Psi}^{n}$ , is a closed, convex subset of  $\mathbb{C}^{n \times n} \times \mathbb{R}$ . Moreover, for  $\Lambda \in \overline{\mathcal{L}_{\Gamma}^{KL}}$ ,  $J_{\Psi}^{n}(\Lambda) < J_{\Psi}^{n+1}(\Lambda)$ . Hence,  $J_{\Psi}^{\infty}$  is well defined and in fact  $J_{\Psi}^{\infty}(\Lambda) = \sup_{n} J_{\Psi}^{n}(\Lambda)$ . It follows that epi $(J_{\Psi}^{\infty}) = \bigcap_{n} \operatorname{epi}(J_{\Psi}^{n})$  is also closed and convex. We conclude that  $J_{\Psi}^{\infty}$  is lower semicontinuous and convex on  $\overline{\mathcal{L}_{\Gamma}^{KL}}$ .

Lemma 2.2.2 Assume that the feasibility condition (1.7) holds. Then,

- 1.  $J_{\Psi}^{\infty}$  is bounded below on  $\overline{\mathcal{L}_{\Gamma}^{KL}}$ ;
- 2.  $J_{\Psi}^{\infty}(\Lambda) = J_{\Psi}(\Lambda)$  on  $\mathcal{L}_{\Gamma}^{KL}$ ;
- 3.  $J_{\Psi}^{\infty}(\Lambda)$  is finite on all of  $\overline{\mathcal{L}_{\Gamma}^{KL}} \setminus \{0\}$ .

*Proof.* By (1.7), there exists  $\Phi_1 \in \mathcal{S}_+(\mathbb{T})$  satisfying (1.9), namely  $\int G\Phi_1 G^* = I$ . Hence, tr( $\Lambda$ ) can be written as tr( $\Lambda \int G\Phi_1 G^*$ ) =  $\int G^* \Lambda G\Phi_1$ , and we get

$$J_{\Psi}^{n}(\Lambda) = \int \left[ G^{*}\Lambda G\Phi_{1} - \Psi \log \left( G^{*}\Lambda G + \frac{1}{n} \right) \right]$$
$$= \int \Phi_{1} \left[ G^{*}\Lambda G - \frac{\Psi}{\Phi_{1}} \log \left( G^{*}\Lambda G + \frac{1}{n} \right) \right].$$

Since the function  $x - \beta \log(x + \frac{1}{n})$  with  $\beta > 0$  attains its minimum at  $x = \beta - \frac{1}{n}$ , we get

$$J_{\Psi}^{n}(\Lambda) = \int \Phi_{1} \left[ G^{*}\Lambda G - \frac{\Psi}{\Phi_{1}} \log \left( G^{*}\Lambda G + \frac{1}{n} \right) \right] \ge \int \psi - \frac{1}{n} \int \Phi_{1} - \mathbb{D}(\Psi || \Phi_{1}).$$

We conclude that  $J_{\Psi}^{\infty} \geq \int \psi - \mathbb{D}(\Psi || \Phi_1)$  on all of  $\overline{\mathcal{L}_{\Gamma}^{KL}}$ . To establish 2, notice that, by Beppo Levi's theorem,

$$J_{\Psi}^{\infty}(\Lambda) := \operatorname{tr}(\Lambda) - \int \lim_{n \to \infty} \Psi \log \left( G^* \Lambda G + \frac{1}{n} \right), \quad \Lambda \in \overline{\mathcal{L}_{\Gamma}^{KL}}.$$
 (2.15)

To prove 3, observe that for  $0 \neq \Lambda \in \partial \mathcal{L}_{\Gamma}^{KL}$ , the boundary of  $\mathcal{L}_{\Gamma}^{KL}$ ,  $G^* \bar{\Lambda} G$  is a nonzero rational spectral density so that  $\log G^* \bar{\Lambda} G$  is integrable over  $\mathbb{T}$  [52, pag. 64]. Since  $\Psi$  is bounded, also  $\Psi \log G^* \bar{\Lambda} G$  is integrable.

In view of these lemmata, we extend  $J_{\Psi}(\Lambda)$  to all of  $\overline{\mathcal{L}_{\Gamma}^{KL}}$  by setting  $J_{\Psi}(\Lambda) := J_{\Psi}^{\infty}(\Lambda)$ on  $\partial \mathcal{L}_{\Gamma}^{KL}$ . Notice that, by (2.15),  $J_{\Psi}$  is finite and given by (2.9) on  $\overline{\mathcal{L}_{\Gamma}^{KL}} \setminus \{0\}$ , and it is  $+\infty$  in  $\Lambda = 0$ .

**Lemma 2.2.3** Assume that the feasibility condition (1.7) holds. Then

$$\lim_{\|\Lambda\| \to +\infty} J_{\Psi}(\Lambda) = +\infty.$$
(2.16)

*Proof.* Recall that by (1.7), there exists  $\Phi_1 \in \mathcal{S}_+(\mathbb{T})$  satisfying (1.9), and, consequently,  $\operatorname{tr}(\Lambda) = \int G^* \Lambda G \Phi_1 > 0, \forall \Lambda \in \mathcal{L}_+$ . Suppose  $\Lambda_k$  is a sequence of matrices in  $\mathcal{L}_{\Gamma}^{KL}$  such that  $\lim_{k\to\infty} ||\Lambda_k|| = +\infty$ . Define the normalized sequence  $\Lambda_k^0 := \frac{\Lambda_k}{||\Lambda_k||}$  (of course, we can assume  $\Lambda_k \neq 0, \forall k$ ). Since  $\operatorname{tr} \Lambda_k^0 > 0$ ,

$$\eta := \liminf_{k \to +\infty} \operatorname{tr} \Lambda_k^0 \ge 0.$$

Consider a sub-sequence such that the limit of its trace is  $\eta$ . This subsequence contains a convergent sub-subsequence  $\{\Lambda_{k_m}^0\}$  since  $\Lambda_k^0$  belongs to the surface of the unit ball, which is compact. Let  $\Lambda_{\infty} := \lim_{m \to \infty} \Lambda_{k_m}^0$ . Since  $G^* \Lambda_n^0 G > 0$  on  $\mathbb{T}$ ,  $G^* \Lambda_{\infty} G \ge 0$  on  $\mathbb{T}$ . Moreover,  $\Lambda_{\infty} \in \text{Range } \Gamma$ , since Range  $\Gamma$  is finite-dimensional, and hence closed. This implies that  $G^* \Lambda_{\infty} G$  cannot be identically zero. In fact, if so,  $\Lambda_{\infty} \in \mathcal{L}_{\Gamma}^{KL} = \text{Range } \Gamma^{\perp}$ . Then  $\Lambda_{\infty} \in \text{Range } \Gamma \cap \text{Range } \Gamma^{\perp} = \{0\}$ , which is a contradiction since  $\|\Lambda_{\infty}\| = 1$ . Thus

$$\eta = \lim_{n \to \infty} \operatorname{tr} \Lambda_n^0 = \operatorname{tr} \Lambda_\infty = \int G^* \Lambda_\infty G \Phi_1 > 0 \tag{2.17}$$

Hence, there exists a K such that  $\operatorname{tr} \Lambda_k^0 > \eta/2$  for all  $k \geq K$ . Finally, since  $G^* \Lambda_k^0 G \leq$ 

 $G^*G$ , we obtain:

$$\begin{split} \liminf_{k \to \infty} J_{\Psi}(\Lambda_k) &= \liminf_{k \to \infty} ||\Lambda_k|| \operatorname{tr} \Lambda_k^0 - \int \Psi \log ||\Lambda_k|| G^* \Lambda_k^0 G \\ &= \liminf_{k \to \infty} ||\Lambda_k|| \operatorname{tr} \Lambda_k^0 - (f \, \Psi) \log ||\Lambda_k|| - \int \Psi \log G^* \Lambda_k^0 G \\ &\geq \liminf_{k \to \infty} ||\Lambda_k|| \operatorname{tr} \Lambda_k^0 - (f \, \Psi) \log ||\Lambda_k|| - \int \Psi \log G^* G \\ &\geq \liminf_{k \to \infty} ||\Lambda_k|| \frac{\eta}{2} - (f \, \Psi) \log ||\Lambda_k|| - \int \Psi \log G^* G \\ &= \liminf_{k \to \infty} \frac{\eta}{2} \left( ||\Lambda_k|| - \frac{\int \Psi}{\eta/2} \log ||\Lambda_k|| \right) - \int \Psi \log G^* G \\ &= +\infty. \end{split}$$

**Theorem 2.2.4** Assume that the feasibility condition (1.7) is satisfied. Then the problem of minimizing the functional  $J_{\Psi}(\Lambda) = \operatorname{tr} \Lambda - \int \Psi \log G^* \Lambda G$  over  $\mathcal{L}_{\Gamma}^{KL}$  admits a unique solution  $\hat{\Lambda} \in \mathcal{L}_{\Gamma}^{KL}$ .

Proof. In view of Lemma 2.2.1, Lemma 2.2.2 and Lemma 2.2.3, the functional  $J_{\Psi}$  is inf-compact on the closed set  $\mathcal{L}_{\Gamma}^{KL}$ , and therefore it admits a minimum point  $\hat{\Lambda}$  there. We show next that  $\hat{\Lambda} \in \mathcal{L}_{\Gamma}^{KL}$ . Of course,  $\hat{\Lambda}$  is not the zero matrix since  $J_{\Psi}(0) = +\infty$ . Let  $0 \neq \overline{\Lambda} \in \partial \mathcal{L}_{\Gamma}^{KL}$ . By Lemma 2.2.2,  $J_{\Psi}(\overline{\Lambda})$  is finite. Observe that, by (1.7),  $I \in \mathcal{L}_{\Gamma}^{KL}$ . By convexity of  $\overline{\mathcal{L}_{\Gamma}^{KL}}$ , it then follows that  $\overline{\Lambda} + \epsilon(I - \overline{\Lambda}) \in \overline{\mathcal{L}_{\Gamma}^{KL}}$ ,  $\forall \epsilon \in [0, 1]$ . We compute the one-sided directional derivative or hemidifferential

$$J'_{\Psi_{+}}(\overline{\Lambda}; I - \overline{\Lambda}) := \lim_{\epsilon \searrow 0} \left[ \frac{J_{\Psi}(\overline{\Lambda} + \epsilon(I - \overline{\Lambda})) - J_{\Psi}(\overline{\Lambda})}{\epsilon} \right]$$
  
= tr(I - \overline{\Lambda}) + \int \Psi - \int \frac{G^\*G\Psi}{G^\*\overline{\Lambda}G} = -\infty. (2.18)

Hence,  $\overline{\Lambda}$  cannot be a minimum point. We conclude that  $\hat{\Lambda} \in \mathcal{L}_{\Gamma}^{KL}$ .

### 2.3 The Pavon-Ferrante algorithm

In general, the optimal solution of the dual problem needs to be computed numerically. This is a delicate problem because of the unboundeness of the gradient of  $J_{\Psi}$  at the boundary of  $\mathcal{L}^{KL}$ , see (2.18). The approaches proposed in [34] and references therein involve some preliminary reparametrization of  $\mathcal{L}^{KL}$ , which may imply loss of global convexity.

In [49], a different matricial iterative method was proposed that appears to be very fast and numerically robust. This method does not restrict the search of  $\hat{\Lambda}$  to  $\mathcal{L}_{\Gamma}^{KL}$  and indeed it normally converges to a  $\hat{\Lambda} \notin \text{Range } \Gamma$ . We show below that this method may be viewed as a modified *gradient descent method* with fixed step size. This method is described as follows.

Let

$$\mathcal{M} := \{ M \in \mathcal{L}^{KL} \mid 0 \le M \le I, \ tr[M] = 1 \},$$
(2.19)

$$\mathcal{M}_+ := \{ M \in \mathcal{M} \mid M > 0 \}.$$

$$(2.20)$$

For  $M \in \mathcal{M}$ , define the map  $\Theta$  by

$$\Theta(M) := \int M^{1/2} G\left[\frac{\Psi}{G^* M G}\right] G^* M^{1/2}.$$
(2.21)

**Theorem 2.3.1** [49]. The map  $\Theta$  maps  $\mathcal{M}$  into  $\mathcal{M}$  and  $\mathcal{M}_+$  into  $\mathcal{M}_+$ .

Consider the following iterative algorithm. Algorithm. Let  $M_0 = \frac{1}{n}I$ . Note that  $M_0 \in \mathcal{M}_+$ . Define the sequence  $\{M_k\}_{k=0}^{\infty}$  by

$$M_{k+1} := \Theta(M_k). \tag{2.22}$$

Notice that, by Theorem 2.3.1,  $M_k \in \mathcal{M}_+$  for all k. Moreover, since  $M_k \in \mathcal{M}, \forall k$ , the sequence is bounded. Hence it has at least one accumulation (limit) point in the closure  $\overline{\mathcal{M}}$  of  $\mathcal{M}$ .

**Theorem 2.3.2** Suppose that the sequence  $\{M_k\}_{k=0}^{\infty}$  has a limit  $\hat{M} \in \mathcal{M}_+$ . Then  $\hat{M} \in \mathcal{L}^{KL}$  and satisfies (2.6), and therefore provides the optimal solution of the approximation problem via (2.7).

Notice that even when the sequence generated by (2.22) converges to a singular matrix  $\hat{M} \in \mathcal{M}$ , it is still possible, though not guaranteed, that such a matrix solves the original problem. We next show that the algorithm may be viewed as a modified gradient descent method. To this aim, rewrite (2.22) as

$$M_{k+1} = M_k + M_k^{1/2} \left[ \int \frac{G\Psi G^*}{G^* M_k G} - I \right] M_k^{1/2}.$$
 (2.23)

Proposition 2.3.3 Define

$$\Delta M_k := M_k^{1/2} \left[ \int \frac{G\Psi G^*}{G^* M_k G} - I \right] M_k^{1/2}, \qquad (2.24)$$

so that (2.23) reads  $M_{k+1} = M_k + \Delta M_k$ . Then,  $\Delta M_k$  is a descent direction at  $M_k$  for  $J_{\Psi}$ .

*Proof.* Let

$$\nabla J_{\Psi}(M_k) = I - \int \frac{G\Psi G^*}{G^* M_k G}$$

denote the "gradient" of  $J_{\Psi}$  at  $M_k$ . Then,

$$\langle \nabla J_{\Psi}(M_k), \Delta M_k \rangle = \operatorname{tr} \left( \nabla J_{\Psi}(M_k) \Delta M_k \right) = -\operatorname{tr} \left( M_k^{1/4} \nabla J_{\Psi}(M_k) M_k^{1/4} \right)^2$$

By Theorem2.3.1,  $M_k > 0$ , for all k. It follows that  $\operatorname{tr} (\nabla J_{\Psi}(M_k) \Delta M_k) < 0$ , unless  $\nabla J_{\Psi}(M_k) = 0$  in which case  $M_k$  is a fixed point of the iteration which solves the dual problem by Theorem(2.3.2).

One could implement the matricial iteration as

$$M_{k+1} = M_k + \alpha_k \Delta M_k, \tag{2.25}$$

where  $0 < \alpha_k \leq 1$  is determined through backstepping, see e.g. [4]. Our extensive simulation (see e.g. [49]), however, shows that convergence in fact occur with  $\alpha_k \equiv 1$ . Indeed, the algorithm appears to perform numerically very well. In fact, at each step the integral (2.21) may be computed very precisely and efficiently via a spectral factorization technique that only requires to solve an algebraic Riccati equation and a Lyapunov equation, both of dimension n. We have performed an extensive number of simulations where the sequence generated by (2.22) never failed to converge. In a very small number of cases, we have observed convergence toward a singular matrix which, however, satisfied (2.6), and therefore provided the optimal solution of the approximation problem.

### 2.4 Difficulties in extending to the multivariate case

In this section, we state and derive some results on multivariable spectrum approximation where a "natural" generalization of the scalar Kullback-Leibler is employed. We also point out the difficulties involved in this approach which bring to a sudden stop the variational analysis.

Multivariable Kullback-Leibler approximation has been investigated in [27, 30], whereas [3] deals with the multivariate Nevanlinna-Pick problem. In statistical quantum mechanics, the state of an *n*-level system is represented by a *density matrix*  $\rho$ , namely a Hermitian, positive-semidefinite matrix in  $\mathbb{C}^{n \times n}$  with unit trace [54]. The convex set of density matrices has as extreme points the one dimensional projections. The latter can be identified with the *pure states* of the system  $|\psi\rangle$ , where  $\psi$  is a unit vector in  $\mathbb{C}^n$ , via  $\rho = \langle \psi, \cdot \rangle \psi$ . Quantum analogues of entropy-like functionals have been considered since the early days of quantum mechanics [60]. Recently, renewed interest has originated in Quantum Information applications [48]. The quantum relative entropy between two density matrices is defined by:

$$\mathbb{D}(\rho || \sigma) := \operatorname{tr}(\rho(\log \rho - \log \sigma)). \tag{2.26}$$

Klein's inequality yields that  $\mathbb{D}(\rho||\sigma) \geq 0$ , and  $\mathbb{D}(\rho||\sigma) = 0$  if and only if  $\rho = \sigma$ . Moreover, as in the classical case, the quantum relative entropy is jointly convex in its arguments. We are then led to the following definition: Given  $\Phi$  and  $\Psi$  in  $\mathcal{S}^{m \times m}_{+}(\mathbb{T})$ , the relative entropy  $\mathbb{D}(\Psi||\Phi)$  is given by

$$\mathbb{D}(\Psi||\Phi) = \int \operatorname{tr}\left(\Psi(\log \Psi - \log \Phi)\right).$$
(2.27)

First of all, we need to worry about nonnegativity of  $\mathbb{D}(\Psi || \Phi)$  and whether it is zero iff  $\Psi = \Phi$ .

**Proposition 2.4.1** Let  $\Phi, \Psi \in \mathcal{S}^{m \times m}_{+}(\mathbb{T})$ . Define  $\Psi_1 = \Psi/\operatorname{tr} \Psi$  and  $\Phi_1 = \Phi/\operatorname{tr} \Phi$ . Then

$$\mathbb{D}(\Psi||\Phi) = \mathbb{D}(\operatorname{tr}\Psi||\operatorname{tr}\Phi) + \int (\operatorname{tr}\Psi)\operatorname{tr}(\Psi_1(\log\Psi_1 - \log\Phi_1)). \quad (2.28)$$

It follows that when  $\int \operatorname{tr} \Psi = \int \operatorname{tr} \Phi$ , then  $\mathbb{D}(\Psi || \Phi) \ge 0$ . Moreover,  $\mathbb{D}(\Psi || \Phi) = 0$  if and only if the two spectra coincide.

Proof.

$$\begin{split} \mathbb{D}(\Psi || \Phi) &= \operatorname{tr} \int \Psi \left( \log \Psi - \log \Phi \right) \\ &= \operatorname{tr} \int \operatorname{tr}(\Psi) \Psi_1 \left( \log \operatorname{tr}(\Psi) \Psi_1 - \log \operatorname{tr}(\Phi) \Phi_1 \right) \\ &= \operatorname{tr} \int \operatorname{tr}(\Psi) \Psi_1 \left( (\log \operatorname{tr}(\Psi)) I + \log \Psi_1 \\ &- (\log \operatorname{tr}(\Phi)) I - \log \Phi_1 \right) \\ &= \operatorname{tr} \int \operatorname{tr}(\Psi) \Psi_1 \left( \log \Psi_1 - \log \Phi_1 \right) \\ &+ \operatorname{tr} \int \operatorname{tr}(\Psi) \Psi_1 \left( (\log \operatorname{tr}(\Psi)) - (\log \operatorname{tr}(\Phi)) \right) I \\ &= \int \operatorname{tr}(\Psi) \operatorname{tr} \left( \Psi_1 \left( \log \Psi_1 - \log \Phi_1 \right) \right) \\ &+ \int \operatorname{tr}(\Psi_1) \operatorname{tr} \Psi \log \frac{\operatorname{tr} \Psi}{\operatorname{tr} \Phi} \\ &= \int \operatorname{tr}(\Psi) \operatorname{tr} \left( \Psi_1 \left( \log \Psi_1 - \log \Phi_1 \right) \right) \\ &+ \mathbb{D}(\operatorname{tr} \Psi || \operatorname{tr} \Phi). \end{split}$$

Since  $\operatorname{tr} \Psi_1(e^{i\vartheta}) = \operatorname{tr} \Phi_1(e^{i\vartheta}) = 1, \forall \vartheta \in [-\pi, \pi]$ , it follows from Klein's inequality that  $\operatorname{tr} \Psi_1(e^{i\vartheta}) \left( \log \Psi_1(e^{i\vartheta}) - \log \Phi_1(e^{i\vartheta}) \right) \geq 0, \quad \forall \vartheta.$ 

The latter implies that

$$\int (\operatorname{tr} \Psi) \operatorname{tr} \left( \Psi_1 \left( \log \Psi_1 - \log \Phi_1 \right) \right) \ge 0.$$

When  $\int \operatorname{tr} \Psi = \int \operatorname{tr} \Phi$ , we also have  $\mathbb{D}(\operatorname{tr} \Psi || \operatorname{tr} \Phi) \ge 0$ . Thus, when  $\int \operatorname{tr} \Psi = \int \operatorname{tr} \Phi$ ,  $\mathbb{D}(\Psi || \Phi)$  is the sum of two nonnegative terms and the conclusion follows. Consider again Problem 2.1.1.

Problem 2.4.2 For  $\Psi \in \mathcal{S}^{m \times m}_{+}(\mathbb{T})$ 

minimize 
$$\mathbb{D}(\Psi \| \Phi)$$
 (2.29)

over 
$$\left\{ \Phi \in \mathcal{S}^{m \times m}_{+}(\mathbb{T}) \mid \int G \Phi G^* = I \right\},$$
 (2.30)

where  $\mathbb{D}(\Psi \| \Phi)$  is defined by (2.27). As in the scalar case, an *a posteriori* rescaling of the prior density is in general necessary. In the light of Proposition 2.4.1, if  $\hat{\Phi}$  is the solution of (2.4.2), the new prior is

$$\hat{\Psi} = \frac{\int \operatorname{tr} \hat{\Phi}}{\int \operatorname{tr} \Psi} \Psi.$$

For  $\Lambda \in \mathbb{C}^{n \times n}$  Hermitian such that  $G^* \Lambda G$  is positive definite on all of  $\mathbb{T}$ , define again the Lagrangian

$$L(\Phi, \Lambda) = \mathbb{D}(\Psi \| \Phi) + \operatorname{tr} \left( \Lambda \left( \int G \Phi G^* - I \right) \right)$$
  
=  $\mathbb{D}(\Psi \| \Phi) + \operatorname{tr} \int G^* \Lambda G \Phi - \operatorname{tr}(\Lambda).$  (2.31)

The following step, entailing the unconstrained minimization of the strictly convex functional  $L(\Phi, \Lambda)$  on  $\Psi \in \mathcal{S}^{m \times m}_{+}(\mathbb{T})$ , is a stumbling block. The optimality condition reads [30, Section IV]

$$\int_0^\infty (\hat{\Phi}_{KL} + \tau I)^{-1} \Psi (\hat{\Phi}_{KL} + \tau I)^{-1} d\tau = G^* \Lambda G.$$
 (2.32)

In general, an explicit expression for  $\hat{\Phi}_{KL}$  in terms of  $\Psi$  and  $\Lambda$  cannot be obtained, and the variational analysis ends here. We mention that the minimization with respect to the first argument of the relative entropy can instead be carried out explicitly, leading to a solution of the exponential form

$$\Phi^o = c \exp(\log \Psi - G^* \Lambda G),$$

see [30, Section IV]. Homotopy like methods are described in [30] to find  $\Lambda$ , when it exists, such that  $\Phi^o$  satisfies the constraint.

These difficulties in solving the approximation problem in the multivariable case, with respect to the Kullback-Leibler metric, motivated the search for other distances between spectra that could be regarded as "natural", and in which the variational analysis could be carried through. This lead us to the Hellinger distance, which is described in the next chapter.

### Chapter 3

# Approximation in the Hellinger distance

### 3.1 The Hellinger distance

The Hellinger distance between two probability densities p and q, defined over a set  $X \subset \mathbb{R}$ , is defined as follows:

$$d_H(p,q) := \sqrt{\int_X \left(\sqrt{p(x)} - \sqrt{q(x)}\right)^2 dx}$$

Let us apply the same concept to spectral density functions. Given  $\Phi \ge 0$  and  $\Psi \ge 0$ in  $L^1(\mathbb{T})$ , we define the *Hellinger distance* between them by

$$d_H(\Phi,\Psi) := \left[ \int_{-\pi}^{\pi} \left( \sqrt{\Phi(e^{i\theta})} - \sqrt{\Psi(e^{i\theta})} \right)^2 \frac{d\theta}{2\pi} \right]^{1/2}.$$

The following properties hold:

**Proposition 3.1.1**  $d_H$  is a bona fide distance.

*Proof.* Since  $\Phi$  and  $\Psi$  belong to  $L^1(\mathbb{T})$ ,  $\sqrt{\Phi}$  and  $\sqrt{\Psi}$  belong to  $L^2(\mathbb{T})$ , and  $d_H(\Phi, \Psi)$  is nothing more than the  $L^2$  distance between  $\sqrt{\Phi}$  and  $\sqrt{\Psi}$ . Thus:

- $d_H(\Phi, \Psi) \ge 0$ ,
- $d_H(\Phi, \Psi) = 0$  if and only if  $\Phi = \Psi$  a.e., because the square root is injective,
- the symmetry is obvious, and
- $d_H(\Phi, \Psi) = ||\sqrt{\Phi} \sqrt{\Psi}||_2 \le ||\sqrt{\Phi} \sqrt{\Omega}||_2 + ||\sqrt{\Omega} \sqrt{\Psi}||_2 = d_H(\Phi, \Omega) + d_H(\Omega, \Psi)$ for any  $\Phi, \Psi, \Omega$ , which proves the triangular inequality.

**Proposition 3.1.2** Consider  $\Phi, \Psi \in S_+(\mathbb{T})$ . Then

1. 
$$d_H(\Phi, \Psi) \leq \sqrt{\|\Phi\|_1 + \|\Psi\|_1};$$
  
2.  $d_H(\Phi, \Psi)^2 \leq \|\Phi - \Psi\|_1;$   
3.  $\|\Phi - \Psi\|_1 \leq \left(\sqrt{\|\Phi\|_1} + \sqrt{\|\Psi\|_1}\right) d_H(\Phi, \Psi).$ 

*Proof.* Observe that

$$\int_{-\pi}^{\pi} \left(\sqrt{\Phi} - \sqrt{\Psi}\right)^2 \frac{d\theta}{2\pi} = \int_{-\pi}^{\pi} \left(\Phi + \Psi - 2\sqrt{\Phi\Psi}\right) \frac{d\theta}{2\pi}$$
$$\leq \int_{-\pi}^{\pi} \left(\Phi + \Psi\right) \frac{d\theta}{2\pi}$$

which proves 1). Also

$$\int_{-\pi}^{\pi} |\Phi - \Psi| \frac{d\theta}{2\pi} = \int_{-\pi}^{\pi} |\sqrt{\Phi} - \sqrt{\Psi}| |\sqrt{\Phi} + \sqrt{\Psi}| \frac{d\theta}{2\pi}$$
$$\geq \int_{-\pi}^{\pi} |\sqrt{\Phi} - \sqrt{\Psi}| |\sqrt{\Phi} - \sqrt{\Psi}| \frac{d\theta}{2\pi}$$

which proves 2). Finally

$$\int_{-\pi}^{\pi} |\Phi - \Psi| \frac{d\theta}{2\pi} = \int_{-\pi}^{\pi} |\sqrt{\Phi} - \sqrt{\Psi}| |\sqrt{\Phi} + \sqrt{\Psi}| \frac{d\theta}{2\pi}$$
$$\leq d_H(\Phi, \Psi) \left[ \int_{-\pi}^{\pi} \left( \Phi + \Psi + 2\sqrt{\Phi\Psi} \right) \frac{d\theta}{2\pi} \right]^{1/2}$$
$$\leq d_H(\Phi, \Psi) \left[ \int_{-\pi}^{\pi} (\Phi + \Psi) \frac{d\theta}{2\pi} + 2\sqrt{\int \Phi \int \Psi} \right]^{1/2}$$
$$= \left( \sqrt{\|\Phi\|_1} + \sqrt{\|\Psi\|_1} \right) d_H(\Phi, \Psi),$$

where we have used the Cauchy-Schwarz inequality twice. This establishes 3).

**Remark 3.1.3** On a finite-dimensional statistical manifold, endowed with the Fisher information as the metric tensor, both the Hellinger distance and the Kullback-Leibler pseudo-distance can be viewed as instances of the broader concept of  $\alpha$ -divergences between two points, which arise from the so-called Amari connections. In particular, the 0-divergence, which indeed is the Hellinger distance, arises from the Levi-Civita connection. See [1, p. 66 and following].

### **3.2** Scalar approximation

As was said before, the choice of the Hellinger distance was motivated by the fact that it generalizes nicely to the case when  $\Phi$  and  $\Psi$  are multivariate spectra. More precisely, the scalar distance, the scalar variational analysis, the result of such analysis, and the existence proof for the optimum of the *dual* problem, will all be particular cases of their multivariable counterparts.

Nevertheless, in order to bridge between the scalar Kullback-Leibler and the multivariable Hellinger theories, let us show how to compute the solution to the problem of finding the scalar spectrum  $\Phi$  that is closest to the scalar  $\Psi$  with respect to the classical Hellinger distance we have just introduced.

### 3.2.1 Variational analysis

The fast-paced variational analysis that follows will omit some details, results and proofs which will be given when we will consider the multivariable case. The reader will notice many similarities with the Kullback-Leibler analysis that was shown in Chapter 2.

**Problem 3.2.1** Let  $\Psi \in S_+(\mathbb{T})$ , and suppose that the feasibility condition (1.7) holds (*i.e.*  $I \in \operatorname{Range} \Gamma$ ). Find  $\hat{\Phi}$  that solves

minimize 
$$d_H^2(\Phi, \Psi)$$
  
over  $\left\{ \Phi \in \mathcal{S}_+(\mathbb{T}) \mid \int G \Phi G^* = I \right\},$  (3.1)

Define

$$\mathcal{L}^{H} := \{ \Lambda \in \mathcal{H}(n) \mid 1 + G^* \Lambda G > 0 \; \forall e^{i\theta} \in \mathbb{T} \}.$$

For  $\Lambda \in \mathcal{L}^{H}$ , consider the Lagrangian function

$$L(\Phi, \Lambda) = d_H^2(\Phi, \Psi) + \left\langle \Lambda, \int G \Phi G^* - I \right\rangle$$
  
=  $d_H^2(\Phi, \Psi) + \operatorname{tr}\left(\Lambda \left(\int G \Phi G^* - I\right)\right)$   
=  $d_H^2(\Phi, \Psi) + \int G^* \Lambda G \Phi - \operatorname{tr} \Lambda.$  (3.2)

Next, consider the *unconstrained* minimization of  $L(\Phi, \Lambda)$ .

minimize{
$$L(\Phi, \Lambda) \mid \Phi \in \mathcal{S}_{+}(\mathbb{T})$$
} (3.3)

**Remark 3.2.2** First of all, observe that  $S_+(\mathbb{T})$  is an open, convex set. Second, notice that, for each  $\Lambda$ , the functional

$$\Phi \mapsto L(\Phi, \Lambda)$$

is strictly convex. Thus, (3.3) is a convex optimization problem. Finally, observe that  $L(\Phi, \Lambda)$  Gâteaux is differentiable at  $\Phi$  in any direction  $\delta \Phi \in L^1(\mathbb{T})$ .

It then follows from a basic result of convex optimization that  $\hat{\Phi} \in \mathcal{S}_+(\mathbb{T})$  solves problem (3.3) if and only if it satisfies the condition

$$\delta L(\hat{\Phi}, \Lambda; \delta \Phi) = 0, \quad \forall \delta \Phi \in C(\mathbb{T}).$$
(3.4)

Here,  $\delta L(\hat{\Phi}, \Lambda; \delta \Phi)$ , the first variation of L at  $\hat{\Phi}$  in direction  $\delta \Phi$ , is defined by

$$\delta L(\hat{\Phi}, \Lambda; \delta \Phi) = \lim_{\epsilon \to 0} \frac{1}{\epsilon} \left[ L(\hat{\Phi} + \epsilon \delta \Phi, \Lambda) - L(\hat{\Phi}, \Lambda) \right].$$

**Proposition 3.2.3** The unique solution  $\hat{\Phi}$  to problem (3.3) is given by

$$\hat{\Phi} = \frac{\Psi}{(1 + G^* \Lambda G)^2}.$$
(3.5)

*Proof.* For  $\Phi \in \mathcal{S}_+(\mathbb{T})$ , we get

$$\delta L(\Phi,\Lambda;\delta\Phi) = \int \left(1 - \Psi^{1/2} \Phi^{-1/2} + G^* \Lambda G\right) \delta\Phi$$

By (3.4),  $\hat{\Phi} \in \mathcal{S}_+(\mathbb{T})$  solves (3.3) if and only if

$$\int \left(1 - \Psi^{1/2} \Phi^{-1/2} + G^* \Lambda G\right) \delta \Phi = 0, \quad \forall \delta \Phi \in L^1(\mathbb{T}).$$
(3.6)

We get

$$\hat{\Phi}^{1/2} = \frac{\Psi^{1/2}}{1 + G^* \Lambda G},\tag{3.7}$$

from which (3.5) follows.

In the spirit of Lagrange, we get the following elementary, albeit fundamental, result

**Theorem 3.2.4** Suppose  $\hat{\Lambda} \in \mathcal{L}^H$  is such that

$$\int G \, \frac{\Psi}{(1+G^*\hat{\Lambda}G)^2} \, G^* = I. \tag{3.8}$$

Then  $\hat{\Phi}$  given by

$$\hat{\Phi} = \frac{\Psi}{(1 + G^* \hat{\Lambda} G)^2} \tag{3.9}$$

is the unique solution of the approximation problem (3.2.1). Proof. Let  $\Phi \in S_+(\mathbb{T})$  satisfy the constraint

$$\int G\Phi G^* = I. \tag{3.10}$$

By Proposition (3.2.3), and by the strict convexity of the functional  $L(\cdot, \hat{\Lambda})$ , we get

$$d_H^2(\Phi, \Psi) = L(\Phi, \hat{\Lambda}) > L(\hat{\Phi}, \hat{\Lambda}) = d_H^2(\hat{\Phi}, \Psi).$$

By (3.8),  $\hat{\Phi}$  in (3.9) satisfies the constraint (3.10). Hence it is optimal for the original constrained problem.

Thus, the original problem (3.2.1) is now reduced to finding  $\hat{\Lambda} \in \mathcal{L}^H$  satisfying (3.8). This is accomplished via duality theory.

#### 3.2.2 The dual problem

In view of Proposition 3.2.3, for  $\Lambda \in \mathcal{L}^H$ , the dual functional takes the form

$$\Lambda \mapsto L\left(\frac{\Psi}{(1+G^*\Lambda G)^2},\Lambda\right) = \int \frac{\Psi G^*\Lambda G}{1+G^*\Lambda G} - \operatorname{tr}(\Lambda).$$
(3.11)

We consider the maximization of (3.11) over  $\mathcal{L}^{H}$ . Let

$$J_{\Psi}(\Lambda) := \operatorname{tr} \Lambda - \int \frac{\Psi G^* \Lambda G}{1 + G^* \Lambda G}$$

The dual problem is then equivalent to

minimize 
$$\{J_{\Psi}(\Lambda) \mid \Lambda \in \mathcal{L}^H\}.$$
 (3.12)

**Remark 3.2.5** Notice that  $\mathcal{L}^H$  is convex. Moreover,  $J_{\Psi}(\Lambda)$  is convex on  $\mathcal{L}^H$ , but, in general, not strictly convex.

 $J_{\Psi}$  is indeed strictly convex when restricted to Range  $\Gamma$ . This can be established along the lines of [34, Section V], and will be shown in detail in the treatment of the multivariable case. Observe that  $\mathcal{L}^{H}$  is an open subset of  $\mathcal{H}(n)$  and  $J_{\Psi}$  is Gâteaux differentiable in any direction  $\delta \Lambda \in \mathcal{H}(n)$ .

Hence,  $\hat{\Lambda} \in \mathcal{L}$  solves (3.12) if and only if, for all  $\delta \Lambda \in \mathcal{H}$ , we have

$$\delta J_{\psi}(\hat{\Lambda};\delta\Lambda) = \operatorname{tr}\left[\left(I - \int \frac{G\Psi G^*}{(1 + G^*\hat{\Lambda}G)^2}\right)\delta\Lambda\right] = 0.$$
(3.13)

Arguing as in the minimization of the Lagrangian, we get the following result.

**Proposition 3.2.6**  $\hat{\Lambda} \in \mathcal{L}$  solves (3.12) if and only if it satisfies (3.8).

**Remark 3.2.7** Existence of a minimum of Jpsi on  $\mathcal{L}^H$  may be established along the lines of [34], or in a similar fashion to Section 2.2. Such a  $\hat{\Lambda}$  then provides the optimal solution of the primal problem (3.2.1).

As for the Kullback-Leibler case, a closed form solution of the dual problem may be obtained only in certain specific cases. In general, one needs to resort to an iterative scheme.

### 3.3 A naïve multivariable generalization

Recall that, for a positive semidefinite Hermitian matrix M, the square root  $M^{1/2}$ of M is the unique Hermitian matrix whose square is M. If V is a unitary matrix that diagonalizes M so that  $M = V^* \operatorname{diag}(\alpha_1^2, \ldots, \alpha_m^2)V$ , then simply  $M^{1/2} =$   $V^* \operatorname{diag}(\alpha_1, \ldots, \alpha_m) V$ . Motivated by the analogy with the Kullback-Leibler case, and by the scalar case, we define the Hellinger distance for  $\Phi$  and  $\Psi$  in  $\mathcal{S}^{m \times m}_+(\mathbb{T})$  to be

$$d_{H}^{2}(\Phi,\Psi) := \int_{-\pi}^{\pi} \operatorname{tr} \left[ \Phi^{1/2}(e^{i\vartheta}) - \Psi^{1/2}(e^{i\vartheta}) \right]^{2} \frac{d\vartheta}{2\pi}.$$
 (3.14)

Notice that (3.14) appears also as the natural generalization of the Hellinger distance for density operators of statistical quantum physics introduced in [44]. Consider again the strictly convex Problem 3.2.1:

minimize 
$$d_H^2(\Phi, \Psi)$$
 (3.15)

over 
$$\left\{ \Phi \in \mathcal{S}^{m \times m}_{+}(\mathbb{T}) \mid \int G \Phi G^* = I \right\},$$
 (3.16)

where  $d_H^2(\Phi, \Psi)$  is now given by (3.14). Define  $\mathcal{L}^H$  by

$$\mathcal{L}^{H} := \{ \Lambda \in \mathcal{H}(n) \mid I + G^{*} \Lambda G > 0 \text{ a.e.on} \mathbb{T} \}.$$
(3.17)

For  $\Lambda \in \mathcal{L}^{H}$ , consider the Lagrangian

$$L(\Phi, \Lambda) = d_H^2(\Phi, \Psi) + \operatorname{tr}\left(\Lambda\left(\int G\Phi G^* - I\right)\right).$$

The unconstrained minimization of the strictly convex functional L over  $\Phi \in \mathcal{S}^{m \times m}_{+}(\mathbb{T})$ , however, leads to an optimality condition (expressing the unique optimum  $\hat{\Phi}_{H}$  in terms of  $\Psi$  and  $\Lambda$ ) that does not appear to be useful.

To obtain such optimality condition, we first need an expression for the directional derivative of the matrix square root. More precisely, given  $P = P^* > 0$  let  $S(P) := P^{1/2}$  and  $\delta P = \delta P^*$ : We want to compute

$$\delta S(P, \delta P) := \lim_{\varepsilon \to 0} \frac{(P + \varepsilon \delta P)^{1/2} - P^{1/2}}{\varepsilon}$$

Employing the chain rule, it is easy to see that

$$\delta S(P, \delta P)P^{1/2} + P^{1/2}\delta S(P, \delta P) = \delta P$$

so that

$$\delta S(P, \delta P) = \int_0^\infty \exp(-P^{1/2}t) \delta P \exp(-P^{1/2}t) dt.$$
 (3.18)

Taking (3.18) into account, we get the optimality condition

$$\int_{0}^{\infty} \left[ \exp(-\hat{\Phi}_{H}^{1/2}t) \left( \hat{\Phi}_{H}^{1/2} - \Psi^{1/2} \right) \exp(-\hat{\Phi}_{H}^{1/2}t) \right] dt + \frac{1}{2} G^* \Lambda G = 0.$$
(3.19)

The integral in (3.19) is the unique solution of the Lyapunov equation

$$\hat{\Phi}^{1/2}X + X\hat{\Phi}^{1/2} = \hat{\Phi}^{1/2} - \Psi^{1/2}.$$
(3.20)

Equations (3.19)-(3.20) now yield

$$-\frac{1}{2}\hat{\Phi}^{1/2}(G^*\Lambda G) - \frac{1}{2}(G^*\Lambda G)\hat{\Phi}^{1/2} = \hat{\Phi}^{1/2} - \Psi^{1/2},$$

which in turn gives

$$\hat{\Phi}^{1/2} \left( I + G^* \Lambda G \right) + \left( I + G^* \Lambda G \right) \hat{\Phi}^{1/2} = 2\Psi^{1/2}.$$
(3.21)

Since  $I + G^* \Lambda G > 0$  almost everywhere on  $\mathbb{T}$ , we finally get

$$\hat{\Phi}^{1/2} = 2 \int_0^\infty \exp\left[-(I + G^* \Lambda G)t\right] \Psi^{1/2} \exp\left[-(I + G^* \Lambda G)t\right] dt.$$
(3.22)

The maximization of the dual functional  $\Lambda \mapsto L(\hat{\Phi}, \Lambda)$ , however, appears quite problematic.

### **3.4** Hellinger distance and spectral factorization

This section and the following one are the core of the whole work. We show in this section that it is possible to define a sensible Hellinger distance for matricial functions that leads to a full unraveling of the complexity of the optimization problem. This will be accomplished by connecting this problem to a most classical topic at the hearth of systems and control theory, namely the *spectral factorization problem*.

Let F be a measurable function defined on the unit circle  $\mathbb{T}$  and taking values in  $\mathbb{C}^{m \times p}$ . Then F belongs to the Hilbert space  $L_2^{m \times p}$  if it satisfies  $\int \operatorname{tr}(FF^*) < \infty$ . For F, G in  $L_2^{m \times p}$ , the scalar product is defined by

$$\langle F, G \rangle_2 = \int \operatorname{tr}(FG^*),$$

so that  $||F||_2^2 = \int \operatorname{tr}(FF^*)$ . Let  $\Phi \in \mathcal{S}^{m \times m}_+(\mathbb{T})$ . Then a measurable  $\mathbb{C}^{m \times p}$ -valued function W is called a spectral factor of  $\Phi$  if it satisfies

$$W(e^{i\vartheta})W(e^{i\vartheta})^* = \Phi(e^{i\vartheta}), \text{ a.e. on } \mathbb{T}.$$

Notice that necessarily  $p \geq m$  and  $W(e^{i\vartheta})$  is a.e. full row rank. Moreover, W is bounded on  $\mathbb{T}$ , and therefore it belongs to  $L_2^{m \times p}$ . When p = m,  $W^{-1}$  is also bounded and, consequently,  $W^{-1} \in L_2^{m \times m}$ . Any  $\Phi \in \mathcal{S}_+^{m \times m}(\mathbb{T})$  satisfies the Szegö condition

$$\int_{-\pi}^{\pi} \log \det \Phi(e^{i\vartheta}) \frac{d\vartheta}{2\pi} > -\infty,$$

and admits therefore spectral factors W in  $H_2^{m \times m}$ , namely the Hardy space of functions in  $L_2^{m \times m}$  that possess an analytic extension in |z| > 1, see e.g. [52, 38].

Let  $W_1$  and  $W_2$  be spectral factors of the same  $\Phi \in \mathcal{S}^{m \times m}_+(\mathbb{T})$  with  $W_1$  square. Then trivially  $U := W_1^{-1} W_2$  is a  $m \times p$  all-pass function, i.e

$$U(e^{i\vartheta})U(e^{i\vartheta})^* = I, \quad \forall e^{i\vartheta} \in \mathbb{T}.$$

For  $\Phi, \Psi \in \mathcal{S}^{m \times m}_{+}(\mathbb{T})$ , consider the following function

$$D_{H}(\Phi, \Psi) = \inf \left\{ \|W_{\Psi} - W_{\Phi}\|_{2} : W_{\Psi}, W_{\Phi} \in L_{2}^{m \times m}, \\ W_{\Psi}W_{\Psi}^{*} = \Psi, W_{\Phi}W_{\Phi}^{*} = \Phi \right\}.$$
(3.23)

**Theorem 3.4.1** The following facts hold true:

1. For any square spectral factor  $\overline{W}_{\Psi}$  of  $\Psi$ , we have:

$$D_{H}(\Phi, \Psi) = \inf \{ \|\bar{W}_{\Psi} - W_{\Phi}\|_{2} : W_{\Phi} \in L_{2}^{m \times m}, \\ W_{\Phi}W_{\Phi}^{*} = \Phi \}.$$
(3.24)

2. The infimum in the above equation is a minimum: Indeed the unique spectral factor of  $\Phi$  minimizing (3.24) is given by

$$\hat{W}_{\Phi} := \Phi^{1/2} \left( \Phi^{1/2} \Psi \Phi^{1/2} \right)^{-1/2} \Phi^{1/2} \bar{W}_{\Psi}$$

- 3.  $D_H$  is a *bona fide* distance function.
- 4.  $D_H$  coincides with the Hellinger distance in the scalar case.

Proof.

1) First of all, observe that, once fixed the spectral factor  $\bar{W}_{\Psi}$ , any square spectral factor  $W_{\Psi}$  of  $\Psi$  can be written as  $W_{\Psi} = \bar{W}_{\Psi}U$ , where U is a  $m \times m$  all-pass. Hence,

$$\int \operatorname{tr}(W_{\Psi} - W_{\Phi})(W_{\Psi} - W_{\Phi})^* d\vartheta$$
$$= \int \operatorname{tr}(\bar{W}_{\Psi} - W_{\Phi}U^*)(\bar{W}_{\Psi} - W_{\Phi}U^*)^* d\vartheta.$$

Observe, moreover, that  $W_{\Phi}U^*$  is a square spectral factor of  $\Phi$ , so that (3.24) holds. 2) To show that the infimum in (3.24) is a minimum, notice that (3.24) may be rewritten in the form

$$D_H(\Phi, \Psi)^2 = \inf \left\{ \int \operatorname{tr}(\bar{W}_{\Psi} - \Phi^{1/2}V)(\bar{W}_{\Psi} - \Phi^{1/2}V)^* d\vartheta : \\ V \in L_{\infty}^{m \times m}, \quad VV^* = I \right\}.$$
(3.25)

We shall solve this problem by unconstrained minimization of the Lagrangian

$$L = \int \operatorname{tr}[(\bar{W}_{\Psi} - \Phi^{1/2}V)(\bar{W}_{\Psi} - \Phi^{1/2}V)^* + \Delta(VV^* - I)],$$
where  $\Delta = \Delta^* > 0$ . The first variation of the Lagrangian (at V in direction  $\delta V \in L^{m \times m}_{\infty}$ ) is

$$\delta L(V;\delta V) = \int \operatorname{tr}[(\Delta V - \Phi^{1/2}W_{\Psi})\delta V^* + \delta V(\Delta V - \Phi^{1/2}W_{\Psi})^*]$$

The second variation of the Lagrangian is

$$\delta^2 L(V;\delta V) = 2 \int \operatorname{tr}[\Phi^{1/2} \delta V \delta V^* \Phi^{1/2} + \Delta^{1/2} \delta V \delta V^* \Delta^{1/2}].$$

Hence, L is strictly convex and therefore V is a minimizer of the unconstrained minimization problem if and only if:

$$\delta L(V;\delta V) = 0, \qquad \forall \ \delta V. \tag{3.26}$$

Condition (3.26) is clearly equivalent to  $\Delta V - \Phi^{1/2} W_{\Psi} = 0$  or to

$$V = \Delta^{-1} \Phi^{1/2} W_{\Psi}.$$

Thus, if there exists  $\Delta = \Delta^* > 0$  such that

$$VV^* = \Delta^{-1} \Phi^{1/2} \Psi \Phi^{1/2} \Delta^{-1} = I,$$

then V minimizes (3.25). Such a  $\Delta$  is readily seen to be given by

$$\Delta = [\Phi^{1/2} \Psi \Phi^{1/2}]^{1/2}$$

In conclusion, the infimum in (3.24) is a minimum and

$$\hat{W}_{\Phi} = \Phi^{1/2} \hat{V} = \Phi^{1/2} [\Phi^{1/2} \Psi \Phi^{1/2}]^{-1/2} \Phi^{1/2} \bar{W}_{\Psi}$$

is the unique minimizer.

3) The distance properties of  $D_H$  are easy to check: (i) Symmetry is an immediate consequence of the definition of  $D_H$ . (ii) It is clear that  $D_H(\Phi, \Phi) = 0$ . Conversely, if  $D_H(\Phi, \Psi) = 0$ , then  $\Phi$  and  $\Psi$  share a.e. a common spectral factor and are, therefore, a.e. the same spectral density. (iii) The triangular inequality is inherited by the definition of  $D_H$  as the infimum of the  $L_2$  distance among spectral factors. Thus, given  $\Phi, \Psi$  and  $\Upsilon$  and chosen an arbitrary square spectral factor  $W_{\Upsilon}$  of  $\Upsilon$ , we have

$$D_{H}(\Phi, \Psi) = \inf_{W_{\Phi}, W_{\Psi}} \|W_{\Phi} - W_{\Psi}\|_{2}$$

$$\leq \inf_{W_{\Phi}, W_{\Psi}} [\|W_{\Phi} - W_{\Upsilon}\|_{2} + \|W_{\Psi} - W_{\Upsilon}\|_{2}]$$

$$= \inf_{W_{\Phi}} \|W_{\Phi} - W_{\Upsilon}\|_{2} + \inf_{W_{\Psi}} \|W_{\Psi} - W_{\Upsilon}\|_{2}$$

$$= D_{H}(\Phi, \Upsilon) + D_{H}(\Psi, \Upsilon)$$

where the last equality is a consequence of point 1).

4) By choosing  $\overline{W}_{\Psi} = \Psi^{1/2}$ , it is immediate to check that in the scalar case (m = 1) $\widehat{V} \equiv 1$  and hence  $D_H$  coincides with the Hellinger distance  $d_H$ .

## **3.5** $D_H$ -optimal multivariable spectrum approximation

Theorem 3.4.1 shows that  $D_H$  is a natural extension to the multivariable case of the Hellinger distance. The corresponding multivariable version of Problem 3.2.1 is the following:

**Problem 3.5.1** Given  $\Psi \in \mathcal{S}^{m \times m}_{+}(\mathbb{T})$ , find  $\hat{\Phi} \in \mathcal{S}^{m \times m}_{+}(\mathbb{T})$  that solves

minimize 
$$D_H^2(\Phi, \Psi)$$
 (3.27)

subject to 
$$\int G\Phi G^* = I.$$
 (3.28)

It is in this form that the optimization problem is amenable to the variational analysis even in multivariable version. Let

$$\mathcal{L}^{H} := \{ \Lambda \in \mathcal{H}(n) \mid I + G^* \Lambda G > 0 \; \forall e^{i\vartheta} \in \mathbb{T} \},$$
(3.29)

and

$$\mathcal{L}_{\Gamma}^{H} := \mathcal{L}^{H} \cap \operatorname{Range} \Gamma \tag{3.30}$$

The following is our main result.

**Theorem 3.5.2** Assume condition (1.8) (or, equivalently, condition (1.10)) is satisfied. Then there exists a unique  $\hat{\Lambda} \in \mathcal{L}_{\Gamma}^{H}$  such that

$$\int G(I + G^* \hat{\Lambda} G)^{-1} \Psi(I + G^* \hat{\Lambda} G)^{-1} G^* = I.$$
(3.31)

The unique solution of the constrained approximation Problem 3.5.1 is then given by

$$\hat{\Phi}_H := (I + G^* \hat{\Lambda} G)^{-1} \Psi (I + G^* \hat{\Lambda} G)^{-1}.$$
(3.32)

**Remark 3.5.3** Let  $\Psi_0 \in S_+(\mathbb{T})$  and suppose  $\Psi = \Psi_0 I$  has the form of a scalar matrix. Then, a simple calculation shows that (3.32) and (3.22) give the same form for the optimal solution  $\hat{\Phi}$ .

We break the proof of Theorem 3.5.2 into three parts: First, by unconstrained minimization of the Lagrangian function, we obtain an expression for a spectral factor of the optimal  $\Phi$  depending on the Lagrange multiplier matrix  $\Lambda$  (Lemma 3.5.5). Second, we establish some regularity properties of the functional  $J_{\Psi}$ , in particular its strict convexity (Theorem 3.5.8). Third, we establish existence of a unique  $\Lambda \in \mathcal{L}_{\Gamma}^{H}$ satisfying (3.31) (Theorem3.5.10).

We begin by recalling a few basic definitions and facts from multivariate analysis. A function  $f: S \subset \mathbb{R}^N \to \mathbb{R}^M$  is (*Fréchet*) differentiable on the open set S if for all  $x \in S$  there exists a linear map  $L_x: \mathbb{R}^N \to \mathbb{R}^M$  such that

$$\lim_{h \to 0} \frac{||f(x+h) - f(x) - L_x(h)||}{||h||} = 0.$$

A function f is said to be  $C^0(S)$  if it is continuous on S. Also, f is said to be  $C^1(S)$  if it is differentiable at each  $x \in S$  and if the operator Df defined by

$$Df(x) := L_x$$

is  $C^0(S)$ . Now the derivative  $Df: S \to L(\mathbb{R}^N, \mathbb{R}^M) \simeq \mathbb{R}^{MN}$  is itself a function between finite-dimensional spaces. If Df is  $C^1(S)$ , then f is said to be  $C^2(S)$ . Proceeding in this way, the  $C^k$ -differentiability of f can be defined. Finally, f is said to be of class  $C^{\infty}(S)$  if it is  $C^k(S)$  for all  $k \in \mathbb{N}$ . A standard result in analysis states that  $f \in C^1(S)$  if and only if the partial derivatives  $\frac{\partial f_m}{\partial x_n}$  (where  $f_m$  is the m-th component of f) exist and are continuous on S (see for instance [53, Theorem 9.21]). It follows that  $f \in C^k(S)$  if and only if f has in S continuous partial derivatives of any order up to k, that is:

$$\frac{\partial^h f_m}{\partial x_{n_1} \cdots \partial x_{n_h}} \in C^0(S)$$

for all  $m, n_i, h$  s.t.  $1 \le m \le M, 1 \le n_i \le N$ , and  $0 \le h \le k$ .

**Remark 3.5.4** In the following, we will deal with functionals, such as the Lagrangian and (the opposite of) the dual functional  $J_{\Psi} : \mathcal{L}_{\Gamma}^{H} \to \mathbb{R}$ . In particular, with respect to  $J_{\Psi}$  the role of the above partial derivatives is played by the directional (or Gâteaux) derivatives,

$$\delta^h J_{\Psi}(\Lambda; \bar{\delta\Lambda}_{n_1}, ..., \bar{\delta\Lambda}_{n_h})$$

where  $1 \leq n_i \leq d$  and  $\{\delta \Lambda_1, ..., \delta \Lambda_d\}$  is a fixed orthonormal base of Range  $\Gamma$ . A fortiori, if we show that  $J_{\Psi}$  has on  $\mathcal{L}_{\Gamma}^H$  continuous directional derivatives of any order up to k, taken in whatever directions  $\{\delta \Lambda_1, ..., \delta \Lambda_k\} \subset \text{Range } \Gamma$ , then we can say that  $J_{\Psi} \in C^k(\mathcal{L}_{\Gamma}^H)$ .

Finally, a standard result in the theory of convex functions states that if a function  $f: S \subset \mathbb{R}^N \to \mathbb{R}$  is  $C^2(S)$  (where S is open), then f is *strictly convex* on S if and only if its Hessian  $H_x$  is positive definite at each  $x \in S$ .

For  $\Lambda \in \mathcal{L}^{H}$ ,  $W_{\Psi}$  a spectral factor of  $\Psi$ , and  $W, W^{-1} \in L^{m \times m}_{\infty}(\mathbb{T})$ , form the Lagrangian function:

$$L(W,\Lambda) = \operatorname{tr} \int (W - W_{\Psi}) (W - W_{\Psi})^{*} + \operatorname{tr} \Lambda \left( \int GWW^{*}G^{*} - I \right).$$

$$(3.33)$$

Consider the unconstrained minimization problem:

$$\min_{W} \left\{ L(W,\Lambda) \mid W, W^{-1} \in L^{m \times m}_{\infty}(\mathbb{T}) \right\}$$
(3.34)

**Lemma 3.5.5** The unique solution to problem (3.34) is given by

$$\hat{W} = (I + G^* \Lambda G)^{-1} W_{\Psi}.$$
(3.35)

*Proof.* The first variation (or *directional derivative*) of the Lagrangian in an arbitrary direction  $\delta W_1 \in L^{m \times m}_{\infty}(\mathbb{T})$  is:

$$\delta L(W,\Lambda;\delta W_1) = \operatorname{tr} \int \left[ \delta W_1 \left( W - W_{\Psi} \right)^* + \left( W - W_{\Psi} \right) \delta W_1^* \right. \\ \left. + \Lambda \left( G \delta W_1 W^* G^* + G W \delta W_1^* G^* \right) \right] \\ = \operatorname{tr} \int \left( W - W_{\Psi} + G^* \Lambda G W \right) \delta W_1^* \\ \left. + \left( \operatorname{tr} \int \left( W - W_{\Psi} + G^* \Lambda G W \right) \delta W_1^* \right)^* \right.$$
(3.36)

As W varies in a bounded subset of  $L^{m \times m}_{\infty}(\mathbb{T})$ ,  $\delta L(W, \Lambda; \delta W_1)$  is a bounded quantity. As a function of W,  $\delta L(W, \Lambda; \delta W_1)$  is therefore the sum of a constant term  $(-\operatorname{tr} \int \delta W_1 W^*_{\Psi} + W_{\Psi} \delta W^*_1)$  and of a bounded linear functional. Hence, in the variable W it is a *continuous* functional. By taking into account the cyclic property of the trace operator, the second variation of the Lagrangian in an arbitrary direction  $\delta W_2 \in L^{m \times m}_{\infty}(\mathbb{T})$  is easily seen to be given by

$$\delta^2 L(W,\Lambda;\delta W_1,\delta W_2) = \operatorname{tr} \int \delta W_1^* \left(I + G^*\Lambda G\right) \delta W_2 + \operatorname{tr} \int \delta W_2^* \left(I + G^*\Lambda G\right) \delta W_1$$
(3.37)

This second variation does not depend on W, hence it is trivially continuous as a function of W. In particular, the second variation taken two times in the same direction  $\delta W$  is:

$$\delta^2 L(W,\Lambda;\delta W) = 2\operatorname{tr} \int \delta W^* \left(I + G^*\Lambda G\right) \delta W$$
(3.38)

which is clearly positive for any  $\Lambda \in \mathcal{L}^H$  and  $\delta W \neq 0$ . Resuming, on any finitedimensional subspace of  $L_{\infty}^{m \times m}(\mathbb{T})$ , both  $\delta L(W, \Lambda; \delta W_1)$  (the "gradient" of L) and  $\delta^2 L(W, \Lambda; \delta W_1, \delta W_2)$  (the "Hessian" of L) are continuous in the variable W, and the second is a positive definite bilinear form. Hence L is strictly convex with respect to W on any finite-dimensional subspace of  $L_{\infty}^{m \times m}(\mathbb{T})$ , and therefore on  $L_{\infty}^{m \times m}(\mathbb{T})$  itself, which implies that any minimum is unique. Finally, the set  $\mathcal{L}^H$  is open and convex. To find the minimum point of L, we impose  $\delta L(W, \Lambda; \delta W) = 0$  in each direction  $\delta W$ . This yields (3.35). We now consider the question of existence of a matrix  $\hat{\Lambda} \in \mathcal{L}^H$  satisfying (3.31). To this end, we introduce the dual functional

$$L(\hat{W}, \Lambda) = \operatorname{tr} \int \left( (I + G^* \Lambda G)^{-1} W_{\Psi} - W_{\Psi} \right) \times \\ \times \left( (I + G^* \Lambda G)^{-1} W_{\Psi} - W_{\Psi} \right)^* \\ + \operatorname{tr} \left[ \Lambda \left( \int G (I + G^* \Lambda G)^{-1} W_{\Psi} \times \right) \\ \times W_{\Psi}^* (I + G^* \Lambda G)^{-1} G^* - I \right] \\ = \operatorname{tr} \int \Psi - (I + G^* \Lambda G)^{-1} \Psi - \operatorname{tr} \Lambda, \quad \Lambda \in \mathcal{L}^H.$$

$$(3.39)$$

Instead of maximizing (3.39), we consider the equivalent problem of minimizing the functional:

$$J_{\Psi}(\Lambda) := -L(\hat{W}, \Lambda) + \operatorname{tr} \int \Psi$$
  
=  $\operatorname{tr} \int (I + G^* \Lambda G)^{-1} \Psi + \operatorname{tr} \Lambda, \qquad \Lambda \in \mathcal{L}^H.$  (3.40)

In order to establish the fundamental properties of  $J_{\Psi}$ , we need two other lemmas.

**Lemma 3.5.6** Let  $H \in \mathcal{H}(n)$  and m be its minimum eigenvalue. The map  $H \mapsto m$  is continuous.

*Proof.* See Appendix A, Lemma A.0.1.

**Lemma 3.5.7** Define  $Q_{\Lambda}(z) = I + G^*(z)\Lambda G(z)$ . Consider a sequence  $\Lambda_n \in \mathcal{L}_{\Gamma}^H$ converging to  $\Lambda \in \mathcal{L}_{\Gamma}^H$ . Then  $Q_{\Lambda_n}^{-1}$  are well defined and continuous on  $\mathbb{T}$  and converge uniformly to  $Q_{\Lambda}^{-1}$  on  $\mathbb{T}$ .

Proof. Observe that, for  $\Lambda \in \mathcal{L}_{\Gamma}^{H}$ ,  $Q_{\Lambda}$  is a positive definite, continuous matrix function on  $\mathbb{T}$ . By Lemma 3.5.6, there exists a continuous function  $m_{\Lambda}(e^{j\vartheta}) > 0$  such that  $Q_{\Lambda}(e^{j\vartheta}) \geq m_{\Lambda}(e^{j\vartheta})I$  for each  $\vartheta$ . Hence,  $Q_{\Lambda}(e^{j\vartheta}) \geq m_{\Lambda}I, \forall \vartheta$ , where  $m_{\Lambda} := \min_{\vartheta} m_{\Lambda}(e^{j\vartheta}) > 0$ . Let  $\delta\Lambda \in B(0, \varepsilon)$ , the *closed* ball of radius  $\varepsilon$  centered in 0. Now,

$$||G^*(\mathbf{e}^{\mathbf{j}\vartheta})\delta\Lambda G(\mathbf{e}^{\mathbf{j}\vartheta})|| \le ||\delta\Lambda||M_G \le \varepsilon M_G$$

where

$$M_G = \max_{\mathbf{q}} ||G^*(\mathbf{e}^{\mathbf{j}\vartheta})|| ||G(\mathbf{e}^{\mathbf{j}\vartheta})||$$

Thus, if we choose  $\varepsilon < m_{\Lambda}/M_G$ , then  $||G^*(e^{j\vartheta})\delta\Lambda G(e^{j\vartheta})|| < m_{\Lambda}$ . Hence,  $I + G^*(\Lambda + \delta\Lambda)G$  describes, as  $(\delta\Lambda, \vartheta)$  varies in  $B(0, \varepsilon) \times [-\pi, \pi]$ , a compact set that does not

contain any singular matrix. Now recall that the matrix inversion operator is continuous at any nonsingular matrix. Hence,  $Q_{\Lambda+\delta\Lambda}^{-1}(e^{j\vartheta})$  admits a uniform bound  $M(\Lambda,\epsilon)$ on  $B(0,\varepsilon) \times [-\pi,\pi]$ . Since  $\Lambda_n \to \Lambda$ , for *n* sufficiently large,  $(\Lambda_n - \Lambda) \in B(0,\varepsilon)$ . Then

$$\sup_{\vartheta} \|Q_{\Lambda_n}^{-1} - Q_{\Lambda}^{-1}\| = \sup_{\vartheta} \|Q_{\Lambda_n}^{-1} [G^*(\Lambda - \Lambda_n)G] Q_{\Lambda}^{-1}\|$$
$$\leq M^2 \sup_{\vartheta} \|G^*(\Lambda - \Lambda_n)G\|$$
$$\leq M^2 \varepsilon M_G.$$

This implies that  $Q_{\Lambda_n}^{-1} \to Q_{\Lambda}^{-1}$  uniformly on  $\mathbb{T}$ .

**Theorem 3.5.8** Consider  $J_{\Psi} : \mathcal{L}_{\Gamma}^{H} \subset \operatorname{Range} \Gamma \to \mathbb{R}$ . Then

1. 
$$J_{\Psi} \in C^{\infty}(\mathcal{L}_{\Gamma}^{H}).$$

2.  $J_{\Psi}$  is strictly convex on  $\mathcal{L}_{\Gamma}^{H}$ .

#### Proof.

Let  $\mathcal{I}: A \mapsto A^{-1}$  be the matrix inversion operator. Making use of

$$\delta \mathcal{I}(A; \delta A) = -A^{-1} \delta A A^{-1}, \qquad (3.41)$$

the first variation of  $J_{\Psi}(\Lambda)$  in an arbitrary direction  $\delta \Lambda_1$  is found to be:

$$\delta J_{\Psi}(\Lambda; \delta \Lambda_{1}) = -\operatorname{tr} \int Q_{\Lambda}^{-1} G^{*} \delta \Lambda_{1} G Q_{\Lambda}^{-1} \Psi + \operatorname{tr} \delta \Lambda_{1}$$

$$= \left\langle I - \int G Q_{\Lambda}^{-1} \Psi Q_{\Lambda}^{-1} G^{*}, \quad \delta \Lambda_{1} \right\rangle$$
(3.42)

The linear functional  $\nabla J_{\Psi,\Lambda}(\cdot) := \delta J_{\Psi}(\Lambda; \cdot)$  defined by (3.42) is the gradient of  $J_{\Psi}$  at  $\Lambda$ . To prove that  $J_{\Psi} \in C^1(\mathcal{L}_{\Gamma}^H)$  we must show that, for any fixed  $\delta \Lambda_1$ ,  $\delta J_{\Psi}(\Lambda; \delta \Lambda_1)$  is continuous in the variable  $\Lambda$  (it follows that  $\nabla J_{\Psi,\Lambda}(\cdot)$  is also continuous in  $\Lambda$ ). Consider a sequence  $M_n \in \text{Range } \Gamma$  converging to 0. By Lemma 3.5.7,  $Q_{\Lambda+M_n}^{-1}$  converge uniformly to  $Q_{\Lambda}^{-1}$ . Recall that  $\Psi$  is bounded. Applying elementwise the bounded convergence theorem, we get

$$\lim_{n \to +\infty} \int Q_{\Lambda+M_n}^{-1} G^* \delta \Lambda_1 G Q_{\Lambda+M_n}^{-1} \Psi = \int Q_{\Lambda}^{-1} G^* \delta \Lambda_1 G Q_{\Lambda}^{-1} \Psi$$

Hence, for all  $\delta \Lambda_1 \in \text{Range } \Gamma$ ,  $\delta J_{\Psi}(\Lambda; \delta \Lambda_1)$  is continuous, i.e.  $J_{\Psi} \in C^1(\mathcal{L}_{\Gamma}^H)$ . The second variation of  $J_{\Psi}$ , say in direction  $\delta \Lambda_2$ , is easily obtained applying (3.41) and the chain rule to (3.42):

$$\delta^{2} J_{\Psi}(\Lambda; \delta\Lambda_{1}, \delta\Lambda_{2}) = \operatorname{tr} \int W_{\Psi}^{*} Q_{\Lambda}^{-1} G^{*} \delta\Lambda_{2} G Q_{\Lambda}^{-1} G^{*} \delta\Lambda_{1} G Q_{\Lambda}^{-1} W_{\Psi}$$

$$+ \operatorname{tr} \int W_{\Psi}^{*} Q_{\Lambda}^{-1} G^{*} \delta\Lambda_{1} G Q_{\Lambda}^{-1} G^{*} \delta\Lambda_{2} G Q_{\Lambda}^{-1} W_{\Psi}$$

$$(3.43)$$

The bilinear form  $H_{\Lambda}(\cdot, \cdot) := \delta^2 J_{\Psi}(\Lambda; \cdot, \cdot)$  is the Hessian of  $J_{\Psi}$  at  $\Lambda$ . Again, continuity of  $\delta^2 J_{\Psi}(\Lambda; \delta \Lambda_1, \delta \Lambda_2)$  can be established by the previous argument in view of Lemma 3.5.7. Similarly, it can be shown that  $J_{\Psi}$  has continuous directional derivatives of any order. Thus  $J_{\Psi} \in C^k(\mathcal{L}_{\Gamma}^H)$  for any k, and the first assertion follows. Finally, we show that  $J_{\Psi}$  is *strictly* convex on  $\mathcal{L}_{\Gamma}^{H}$ . A standard result in the theory of convex functions states that if a function  $f: S \subset \mathbb{R}^{N} \to \mathbb{R}$  is  $C^{2}(S)$  (where S is open), then f is strictly convex on S if and only if its Hessian  $H_x$  is positive definite at each  $x \in S$ . Consider  $H_{\Lambda}(\delta\Lambda, \delta\Lambda) = \delta^2 J_{\Psi}(\Lambda; \delta\Lambda, \delta\Lambda)$  for  $\delta\Lambda \in \operatorname{Range} \Gamma \setminus \{0\}$ . Since the integrand in (3.43) is positive semidefinite, it follows that  $H_{\Lambda}(\delta\Lambda,\delta\Lambda) \geq 0$ . In view of Point 3 in Proposition 1.3.1, the integrand is not identically zero and  $H_{\Lambda}(\delta\Lambda,\delta\Lambda) > 0$ . It follows that  $J_{\Psi}$  is strictly convex.

As an immediate consequence of the above theorem, we have the following corollary.

Corollary 3.5.9 The dual problem

Find 
$$\Lambda \in \mathcal{L}_{\Gamma}^{H}$$
 minizing  $J_{\Psi}(\Lambda)$  (3.44)

admits at most one solution. Moreover, (3.31) is necessary and sufficient for  $\Lambda$  to solve the dual problem (3.44).

We now tackle the existence issue for the dual problem. Although this is a finitedimensional, convex optimization problem, the existence question is quite delicate since the set  $\mathcal{L}_{\Gamma}^{H}$  is open and unbounded. The proof of the following theorem is partially inspired by the proof of the corresponding result for the scalar, Kullback-Leibler case in [12, Section 2].

**Theorem 3.5.10** If Problem 3.5.1 is feasible, i.e. (1.8) (or, equivalently, condition (1.10)) is satisfied, then the dual functional (3.40) has a unique minimum point in  $\mathcal{L}_{\Gamma}^{H}$ .

*Proof.* In view of Corollary 3.5.9, we only need to show that  $J_{\Psi}$  takes a minimum value on  $\mathcal{L}_{\Gamma}^{H}$ . First, we observe that  $J_{\Psi}$  is continuous on its domain. Second, we show that  $J_{\Psi}$  is bounded below on  $\mathcal{L}_{\Gamma}^{H}$ . Indeed, by feasibility, there exists a  $\bar{\Phi} \in \mathcal{S}^{m \times m}_{+}(\mathbb{T})$ such that  $\int G\overline{\Phi}G^* = I$ . Hence, for all  $M \in \mathbb{C}^{n \times n}$ ,  $\int G\overline{\Phi}G^*M = M$ , which implies

$$\operatorname{tr} M = \operatorname{tr} \int \bar{\Phi}^{1/2} G^* M G \bar{\Phi}^{1/2}.$$
 (3.45)

Recalling that, for  $\Lambda \in \mathcal{L}_{\Gamma}^{H}$ ,  $I + G^{*}(e^{i\vartheta})\Lambda G(e^{i\vartheta})$  is positive definite for all  $\vartheta \in [0, 2\pi)$ , and using the monotonicity property of the trace, we get

$$\operatorname{tr} \Lambda = \operatorname{tr} \int \bar{\Phi}^{1/2} G^* \Lambda G \bar{\Phi}^{1/2} > -\operatorname{tr} \int \bar{\Phi}, \quad \forall \Lambda \in \mathcal{L}_{\Gamma}^H.$$
(3.46)

Define  $\bar{f} := -\operatorname{tr} \int \bar{\Phi} < 0$ . We get

$$J_{\Psi}(\Lambda) := \operatorname{tr} \int (I + G^* \Lambda G)^{-1} \Psi + \operatorname{tr} \Lambda$$
  
=  $\operatorname{tr} \int \Psi^{1/2} (I + G^* \Lambda G)^{-1} \Psi^{1/2} + \operatorname{tr} \Lambda$   
>  $\bar{f}, \quad \forall \Lambda \in \mathcal{L}_{\Gamma}^H,$  (3.47)

where we have used tr  $\int \Psi^{1/2} (I + G^* \Lambda G)^{-1} \Psi^{1/2} > 0$ . on  $\mathcal{L}_{\Gamma}^H$ .

Finally, we show that  $J_{\Psi}$  is *inf-compact* i.e. the sub-level sets  $J_{\Psi}^{-1}(-\infty, r]$  are compact. This implies existence of a minimum point. Indeed, observing that  $J_{\Psi}(0) =$ tr  $\int \Psi$ , we can then restrict the search for a minimum point to the compact set  $J_{\Psi}^{-1}(-\infty, \text{tr} \int \Psi]$ . Existence for the latter problem then follows from a version of Weierstrass Theorem since an inf-compact function has closed level sets and is therefore lower semicontinuous [40, p.56]. To prove inf-compactness of  $J_{\Psi}$ , we proceed to show that:

1.

$$\lim_{\Lambda \to \partial \mathcal{L}_{\Gamma}^{H}} J_{\Psi}(\Lambda) = +\infty$$

where  $\partial \mathcal{L}_{\Gamma}^{H}$  denotes the boundary of  $\mathcal{L}_{\Gamma}^{H}$ ;

2.

$$\lim_{\|\Lambda\| \to \infty} J_{\Psi}(\Lambda) = +\infty.$$

To prove property 1), notice that  $\partial \mathcal{L}_{\Gamma}^{H}$  is the set of  $\Lambda$  in Range  $\Gamma$  for which: (i)  $I + G^*\Lambda G$  is positive semidefinite on  $\mathbb{T}$  and (ii)  $\exists \bar{\vartheta}$  s.t.  $I + G^*(e^{i\bar{\vartheta}})\Lambda G(e^{i\bar{\vartheta}})$  is singular. Thus, for  $\Lambda \to \partial \mathcal{L}_{\Gamma}^{H}$ , all the eigenvalues of  $[I + G^*\Lambda G]^{-1}$  are positive on  $\mathbb{T}$  and, at least one of them, has a pole tending to the unit circle  $([I + G^*\Lambda G]$  and  $[I + G^*\Lambda G]^{-1}$  are *rational* matrix functions!). Since  $\Psi$  is fixed and coercive, then also  $\Psi^{1/2}[I + G^*\Lambda G]^{-1}\Psi^{1/2}$  has all eigenvalues positive on  $\mathbb{T}$  and, at least one of them with a pole tending to the unit circle as  $\Lambda \to \partial \mathcal{L}_{\Gamma}^{H}$ . Rewrite now  $J_{\Psi}$ , as in (3.47), in the form  $J_{\Psi} = \operatorname{tr} \int \Psi^{1/2}(I + G^*\Lambda G)^{-1}\Psi^{1/2} + \operatorname{tr} \Lambda$ . Since  $\operatorname{tr} \Lambda$  is bounded below in view of (3.46), we get the conclusion.

Point 2) is more delicate. Let  $\Lambda_k \in \mathcal{L}_{\Gamma}^H$  be a sequence such that  $\lim_{k\to\infty} \|\Lambda_k\| = \infty$ . Let  $\Lambda_k^0 := \frac{\Lambda_k}{\|\Lambda_k\|}$ . It is easy to see that if  $\Lambda \in \mathcal{L}_{\Gamma}^H$ , then  $\alpha \Lambda \in \mathcal{L}_{\Gamma}^H$  for all  $\alpha \in [0, 1]$ . Hence, for sufficiently large k, we have  $\Lambda_k^0 \in \mathcal{L}_{\Gamma}^H$ .

Let  $\eta = \liminf \operatorname{tr} \Lambda_k^0$ . We want to show that  $\eta$  is strictly positive. We first observe that  $\eta \ge 0$ . In fact,  $\operatorname{tr} \Lambda_k^0 = \frac{1}{\|\Lambda_k\|} \operatorname{tr} \Lambda_k > \frac{1}{\|\Lambda_k\|} \overline{f} \to 0$ , where we have used (3.46).

Consider a sub-sequence of  $\Lambda_k^0$  such that the limit of its trace is  $\eta$ . Since this sub-sequence remains on the surface of the unit ball  $\partial \mathcal{B} := \{\Lambda = \Lambda^* : \|\Lambda\| = 1\}$ , which is compact, it has a sub-sub-sequence converging in  $\partial \mathcal{B}$ . Let  $\Lambda_{k_i}^0$  be such a sub-sub-sequence, and let  $\Lambda^{\infty} \in \partial \mathcal{B}$  be its limit. Clearly,

$$\lim_{i \to \infty} \operatorname{tr} \Lambda^0_{k_i} = \operatorname{tr} \Lambda^\infty = \eta.$$
(3.48)

38

We now prove that  $\Lambda^{\infty} \in \mathcal{L}_{\Gamma}^{H}$ . To this aim, notice that  $\Lambda^{\infty}$  is the limit of a sequence in the finite dimensional linear space Range  $\Gamma$  and hence it belongs to the same space Range  $\Gamma$ . It remains to show that  $(I + G^*\Lambda^{\infty}G)$  is positive definite on  $\mathbb{T}$ . Indeed, since (the unnormalized sub-sequence)  $\Lambda_{k_i}$  belongs to  $\mathcal{L}_{\Gamma}^{H}$ , we have that  $(I + G^*\Lambda_{k_i}G)$ is positive definite on  $\mathbb{T}$  so that  $(\frac{1}{\|\Lambda_{k_i}\|}I + G^*\Lambda_{k_i}^{0}G)$  is also positive definite on  $\mathbb{T}$  for each *i*. Taking the limit for  $i \to \infty$ , we get that  $G^*\Lambda^{\infty}G$  is positive semi-definite on  $\mathbb{T}$  so that  $(I + G^*\Lambda^{\infty}G)$  is strictly positive definite on  $\mathbb{T}$ . This proves that  $\Lambda^{\infty} \in \mathcal{L}_{\Gamma}^{H}$ . The latter, together with (3.45) yields

$$\operatorname{tr} \Lambda^{\infty} = \operatorname{tr} \int \bar{\Phi}^{1/2} G^* \Lambda^{\infty} G \bar{\Phi}^{1/2}.$$
(3.49)

As seen before,  $G^*\Lambda^{\infty}G$  is positive semi-definite on  $\mathbb{T}$ . Moreover,  $G^*\Lambda^{\infty}G$  is not identically zero since  $\Lambda^{\infty} \in \text{Range }\Gamma$ , and  $\Lambda^{\infty} \neq 0$  (it is not the zero matrix) since  $\Lambda^{\infty} \in \partial \mathcal{B}$ . We conclude, in view of (3.48) and (3.49), that  $\eta = \text{tr }\Lambda^{\infty} > 0$ .

Finally, we have

$$J_{\Psi}(\Lambda_k) = \operatorname{tr} \int \Psi^{1/2} (I + G^* \Lambda_k G)^{-1} \Psi^{1/2} + \operatorname{tr} \Lambda_k \ge \|\Lambda_k\| \operatorname{tr} \Lambda_k^0.$$
(3.50)

Since  $\|\Lambda_k\| \to \infty$  and  $\liminf \operatorname{tr} \Lambda_k^0 > 0$ , we get

$$\lim_{k \to \infty} J_{\Psi}(\Lambda_k) = +\infty.$$
(3.51)

Let  $\hat{\Lambda} \in \mathcal{L}_{\Gamma}^{H}$  be the unique solution of the dual problem whose existence has just been proven in Theorem 3.5.10. We show below that it also provides via (3.32) the unique solution to the primal problem 3.5.1.

Proof of Theorem 3.5.2: Let  $W_{\Psi}$  be any spectral factor of  $\Psi$ . Let  $\hat{W} = (I + G^* \Lambda G)^{-1} W_{\Psi}$  as in (3.35). Let W, belonging to  $L^{m \times m}_{\infty}(\mathbb{T})$  together with its inverse, satisfy the constraint

$$\int GWW^*G^* = I. \tag{3.52}$$

By Lemma (3.5.5), and by the strict convexity of the functional  $L(\cdot, \hat{\Lambda})$ , we get

$$\|\hat{W} - W_{\Psi}\|_{2}^{2} = L(\hat{W}, \hat{\Lambda}) < L(W, \hat{\Lambda}) = \|W - W_{\Psi}\|_{2}^{2}.$$

Thus,  $\hat{W}$  minimizes the  $L_2$  distance to  $W_{\Psi}$  among W belonging to  $L_{\infty}^{m \times m}(\mathbb{T})$  together with their inverse and satisfying constraint (3.52). Theorem 3.4.1 now shows that  $\hat{\Phi}_H = \hat{W}\hat{W}^*$  (coinciding with  $\hat{\Phi}_H$  in (3.32)), is the unique solution to the multivariate approximation Problem 3.5.1.

**Remark 3.5.11** Consider the important covariance extension problem when, as it is often the case, the process y is real-valued. Then A and B are real matrices and  $\Psi$  is a real spectral density, i.e.  $\Psi(z)$  is real (and symmetric) for all  $z \in \mathbb{T}$ . In this case,  $\hat{\Lambda}$  is a real symmetric matrix.

# Chapter 4

# A matricial Newton algorithm

## 4.1 The Newton algorithm

The Newton algorithm is an iterative procedure for the search of roots of a function or the *minimization* of a functional. With respect to the latter objective, it can be formulated as follows. Let  $f: S \to \mathbb{R}$  be a functional defined over  $S \subset \mathbb{R}^n$ . In order to find an estimate  $\hat{x}$  of a minimum point  $x^*$  of f,

- 1. Make an initial guess  $x_0$ , possibly near the minimum point.
- 2. At each iteration, compute the Newton step

$$\Delta x_i = -H_{x_i}^{-1} \nabla f_{x_i} \tag{4.1}$$

where  $H_{x_i}$  is the Hessian of f at  $x_i$  and  $\nabla f_{x_i}$  is the gradient of f at  $x_i$  (understood as a column vector).

3. Set  $t_i^0 = 1$ , and let  $t_i^{k+1} = t_i^k/2$  until both of the following conditions hold:

$$x_i + t_i^k \Delta x_i \in S \tag{4.2}$$

$$f(x_i + t_i^k \Delta x_i) < f(x_i) + \alpha t_i^k \nabla f_{x_i}^\top \Delta x_i$$
(4.3)

where  $\alpha$  is a real constant,  $0 < \alpha < 1/2$ .

- 4. Set  $x_{i+1} = x_i + t_i^k \Delta x_i$ .
- 5. Repeat steps 2, 3 and 4 until  $|\nabla f_{x_i}| < \varepsilon$ , where  $\varepsilon$  is a (small) tolerance threshold, then set  $\hat{x} = x_i$ .

In its "pure" form, the iteration of the Newton algorithm only consists in step 2, which is indeed its essential part. Step 3 is the so-called *backtracking* procedure. For small t, if f is sufficiently regular, we have  $f(x_i + t\Delta x_i) \simeq f(x_i) + t\nabla f_{x_i}^{\top}\Delta x_i$ . Since  $\nabla f_{x_i}^{\top}\Delta x_i = -\nabla f_{x_i}^{\top}H_{x_i}^{-1}\nabla f_{x_i} < 0$ , condition (4.3) *must* hold for small t, hence step 3

must terminate at some iteration. Since  $\nabla f_{x_i}^{\top} \Delta x_i < 0$ , (4.3) implies  $f(x_i + t_i^k \Delta x_i) < f(x_i)$ . That is,  $\{f(x_i)\}$  is a strictly decreasing sequence.

In essence, the "pure" Newton algorithm works *very* well when the starting point happens to be near the minimum and the function f is there effectively approximated by a quadratic form, but it can suffer from numerical problems when this is not the case. The backtracking line search is a remedy to this drawback; moreover it can be shown that, under certain regularity assuptions on f, which hold in our case (see Section 4.3), after a finite number of iterations step 3 always selects the multiplier t = 1, that is, the full step. During the latter stage, the convergence to the minimizing solution is quadratic, meaning that there exists a constant C such that  $||x_{i+1} - x^*|| \leq C||x_i - x^*||^2$ . This rate of convergence makes the Newton algorithm often preferable over other minimization methods (see [4]).

## 4.2 Hellinger distance minimization with the Newton Algorithm

We must minimize the functional  $J_{\Psi}(\Lambda)$  over the set  $\mathcal{L}_{\Gamma}^{H}$ . As initial condition, we can safely choose 0. Hence, set

$$\Lambda_0 = 0. \tag{4.4}$$

It turns out that, although the problem is finite-dimensional, the *inversion of the Hessian* is more demanding than inverting a matrix. In order to compute the Newton step  $\Delta \Lambda_i$ , we must solve at  $\Lambda_i$  the following linear equation:

$$H_{\Lambda_i}(\Delta\Lambda_i, \cdot) = -\nabla J_{\Psi,\Lambda_i}(\cdot) \tag{4.5}$$

where, once fixed  $\Lambda_i$ ,  $\nabla J_{\Psi,\Lambda_i}(\cdot)$  and  $H_{\Lambda_i}(\cdot, \cdot)$  must be understood as a linear and a bilinear form, defined by (3.42) and (3.43) respectively. Comparing with the above definitions, (4.5) reduces to:

$$\int GQ_{\Lambda_i}^{-1} \left[ (G^*(\Delta\Lambda_i) GQ_{\Lambda_i}^{-1} \Psi) + (G^*(\Delta\Lambda_i) GQ_{\Lambda_i}^{-1} \Psi)^* \right] Q_{\Lambda_i}^{-1} G^*$$

$$= \int GQ_{\Lambda_i}^{-1} \Psi Q_{\Lambda_i}^{-1} G^* - I$$
(4.6)

In principle, equation (4.6) is not difficult to solve. We suggest the following procedure:

• At the beginning of the procedure, take a base  $\{H_1, ..., H_k, ..., H_N\}$  of  $\mathbb{C}^{m \times n}$ .<sup>1</sup> Then compute the solutions  $\{\Sigma_1, ..., \Sigma_k, ..., \Sigma_N\}$  of the following discrete-time Lyapunov equations:

$$\Sigma_k - A\Sigma_k A^* = BH_k + H_k^* B^*$$

$$R: H \mapsto BH + H^*B^*.$$

<sup>&</sup>lt;sup>1</sup>Actually, it suffices to take the  $\{H_k\}$  to be a base of  $\mathbb{C}^{m \times n} \ominus \operatorname{Ker} R$ , where the map R is defined by

As shown before, these solutions generate Range  $\Gamma$ .

- To compute  $\Delta \Lambda_i$  at each step,
  - 1. Compute the integral

$$Y = \int GQ_{\Lambda_i}^{-1} \Psi Q_{\Lambda_i}^{-1} G^* - I$$
 (4.7)

2. For each  $\Sigma_k$  in the precomputed generators, compute the following integral:

$$Y_{k} = \int GQ_{\Lambda_{i}}^{-1} \left[ (G^{*}\Sigma_{k}GQ_{\Lambda_{i}}^{-1}\Psi) + (G^{*}\Sigma_{k}GQ_{\Lambda_{i}}^{-1}\Psi)^{*} \right] Q_{\Lambda_{i}}^{-1}G^{*}$$

$$(4.8)$$

3. Solve, by means of linear algebraic methods (the Moore-Penrose pseudoinverse), the equation

$$\sum_{k} \alpha_k Y_k = Y \tag{4.9}$$

4. By linearity, the solution to (4.6) is

$$\Delta \Lambda_i = \sum_k \alpha_k \Sigma_k. \tag{4.10}$$

It is clear that the real difficulty here is the computation of the integrals (4.7) and (4.8). This task requires extensive use of the following results of linear stochastic systems theory.

**Lemma 4.2.1** Let A be a stability matrix and  $W(z) = C(zI - A)^{-1}B + D$  a minimal realization of a spectral factor of  $\Phi(z)$ . Let  $\Pi$  be the unique solution to the Lyapunov equation

$$\Pi = A\Pi A^* + BB^* \tag{4.11}$$

Then the following hold:

- 1.  $\int_{-\pi}^{\pi} \Phi(\mathrm{e}^{\mathrm{j}\vartheta}) \frac{\mathrm{d}\vartheta}{2\pi} = C \Pi C^* + D D^*.$
- 2.  $Z(z) = C(zI A)^{-1}(A\Pi C^* + BD^*) + \frac{1}{2}(C\Pi C^* + DD^*)$  is a realization of the causal part of  $\Phi(z)$ ; that is, Z(z) is analytic outside the unit circle and  $\Phi(z) = Z(z) + Z^*(z)$ .

**Lemma 4.2.2** Let  $Z(z) = C(zI - A)^{-1}G + \frac{1}{2}\Sigma$  be a minimal realization of the causal part of a spectrum  $\Phi(z)$ . Let  $P_{-}$  be the stabilizing solution of the following Algebraic Riccati Equation (ARE):

$$P = APA^* + (G - APC^*)(\Sigma - CPC^*)^{-1}(G^* - CPA^*)$$
(4.12)

Let moreover  $D = (\Sigma - CP_-C^*)^{1/2}$  and  $B = (G - AP_-C^*)D^{-1}$ . Then  $W(z) = C(zI - A)^{-1}B + D$  is the minimum phase spectral factor of  $\Phi(z)$ ; that is, W(z) is stable and with stable causal inverse, and  $\Phi(z) = W(z)W^*(z)$ .

**Lemma 4.2.3** Let  $F(z) = C(zI - A)^{-1}B + D$  be a square transfer function, where D is invertible. Then

$$F^{-1}(z) = -D^{-1}C\left(zI - (A - BD^{-1}C)\right)^{-1}BD^{-1} + D^{-1}$$
(4.13)

is a realization of its inverse.

**Lemma 4.2.4** For all matrices  $P = P^* \in \mathbb{C}^{n \times n}$  the following identity holds:

$$\begin{bmatrix} B^*(z^{-1}I - A^*)^{-1} & I \end{bmatrix} \begin{bmatrix} A^*PA - P & A^*PB \\ B^*PA & B^*PB \end{bmatrix} \times \\ \times \begin{bmatrix} (zI - A)^{-1}B \\ I \end{bmatrix} = 0$$
(4.14)

**Lemma 4.2.5** Let A be a stability matrix and  $H(z) = C(zI-A)^{-1}B+D$  be a minimal realization. Let P be the solution of the Lyapunov equation

$$P = A^* P A + C^* C. (4.15)$$

Let 
$$\begin{bmatrix} K \\ J \end{bmatrix}$$
 be an ortho-normal basis of ker  $\begin{bmatrix} A^* P^{1/2} & C^* \end{bmatrix}$  i.e.  
 $\begin{bmatrix} A^* P^{1/2} & C^* \end{bmatrix} \begin{bmatrix} K \\ J \end{bmatrix} = 0, \qquad \begin{bmatrix} K^* & J^* \end{bmatrix} \begin{bmatrix} K \\ J \end{bmatrix} = I.$  (4.16)

Let  $G := P^{-1/2}K$  and

$$H_1(z) := (D^*C + B^*PA)(zI - A)^{-1}G + B^*PG + D^*J.$$
(4.17)

Then,  $H^*(z)H(z) = H_1(z)H_1^*(z)$ .

Lemmas 4.2.1, 4.2.2 and 4.2.3 are standard results (see for example [19]). The proof of Lemma 4.2.4 can be found in [14, Appendix A]. The proof of Lemma 4.2.5 can be found in Appendix A, Lemma A.0.2.

**Remark 4.2.6** Lemma 4.2.5 not only gives us a tool to compute a left factor from a right factor of a given spectrum. It also works in the opposite direction. Indeed, let  $W(z) = C(zI - A)^{-1}B + D$  be a minimal realization, and let  $\zeta = z^{-1}$ . Then

$$\begin{split} \Phi(z) &= W(z)W^*(z) \\ &= (C(zI - A)^{-1}B + D)(B^\top (z^{-1}I - A^\top)^{-1}C^\top + D^\top) \\ &= (B^\top (\zeta^{-1}I - A^\top)^{-1}C^\top + D^\top)^\top (B^\top (\zeta I - A^\top)^{-1}C^\top + D^\top) \\ &= (B^\top (\zeta I - A^\top)^{-1}C^\top + D^\top)^* (B^\top (\zeta I - A^\top)^{-1}C^\top + D^\top) \\ &:= H^*(\zeta)H(\zeta) \end{split}$$

Applying Lemma 4.2.5 we can find an  $H_1(\zeta) = H(\zeta I - F)^{-1}G + K$  such that  $H^*(\zeta)H(\zeta) = H_1(\zeta)H_1^*(\zeta)$ . Now turning back to z:

$$\begin{aligned} H_1(\zeta)H_1^*(\zeta) \\ &= (H(\zeta I - F)^{-1}G + K)(G^\top(\zeta^{-1}I - F^\top)^{-1}H^\top + K^\top) \\ &= (G^\top(z^{-1}I - F^\top)^{-1}H^\top + K^\top)^\top(G^\top(zI - F^\top)^{-1}H^\top + K^\top) \\ &= (G^\top(zI - F^\top)^{-1}H^\top + K^\top)^*(G^\top(zI - F^\top)^{-1}H^\top + K^\top) \\ &= W_1^*(z)W_1(z) = \Phi(z) \end{aligned}$$

# 4.2.1 Factorization of $Q_{\Lambda}^{-1}(z)$

The first problem to solve is to obtain a spectral factor of  $Q_{\Lambda}^{-1}(z)$ , where  $Q_{\Lambda}(z) = (I + G^*(z)\Lambda G(z))$ . To this end, note that

$$Q_{\Lambda}(z) = \begin{bmatrix} B^*(z^{-1}I - A^*)^{-1} & I \end{bmatrix} \begin{bmatrix} \Lambda & 0 \\ 0 & I \end{bmatrix} \times \\ \times \begin{bmatrix} (zI - A)^{-1}B \\ I \end{bmatrix}$$
(4.18)

Applying lemma 4.2.4, we can rewrite (4.18) as

$$Q_{\Lambda}(z) = \begin{bmatrix} B^{*}(z^{-1}I - A^{*})^{-1} & I \end{bmatrix} \times \\ \times \begin{bmatrix} A^{*}PA - P + \Lambda & A^{*}PB \\ B^{*}PA & B^{*}PB + I \end{bmatrix} \times \\ \times \begin{bmatrix} (zI - A)^{-1}B \\ I \end{bmatrix}$$
(4.19)

Now, the following linear matrix inequality:

$$\begin{bmatrix} A^*PA - P + \Lambda & A^*PB \\ B^*PA & B^*PB + I \end{bmatrix}$$

$$= \begin{bmatrix} M^* \\ N^* \end{bmatrix} \begin{bmatrix} M & N \end{bmatrix} \ge 0$$
(4.20)

is solvable for  $P = P^* > 0$  if and only if such is the following ARE:

$$P = A^* P A - A^* P B (B^* P B + I)^{-1} B^* P A + \Lambda$$
(4.21)

The stabilizing solution P of (4.21) gives a realization for the square, minimum phase *co-analytic* spectral factor of Q(z). We have:

$$N = N^* = (B^*PB + I)^{1/2}$$

$$M = (B^*PB + I)^{-1/2}B^*PA$$

$$\Delta_{\Lambda}(z) = \begin{bmatrix} M & N \end{bmatrix} \begin{bmatrix} (zI - A)^{-1}B \\ I \end{bmatrix}$$

$$= (B^*PB + I)^{-1/2}B^*PA(zI - A)^{-1}B$$

$$+ (B^*PB + I)^{1/2}$$

$$Q_{\Lambda}(z) = \Delta_{\Lambda}^*(z)\Delta_{\Lambda}(z)$$
(4.22)

and finally  $Q_{\Lambda}^{-1}(z) = \Delta_{\Lambda}^{-1}(z)\Delta_{\Lambda}^{-*}(z)$  where, by means of lemma 4.2.3,

$$\Delta_{\Lambda}^{-1}(z) = -(B^*PB + I)^{-1}B^*PA(zI - \Gamma)^{-1}B \times \times (B^*PB + I)^{-1/2} + (B^*PB + I)^{-1/2} \Gamma = A - B(B^*PB + I)^{-1}B^*PA$$
(4.23)

## 4.2.2 Computation of the integrals in (4.7) and (4.8)

By virtue of Lemma 4.2.5 and Remark 4.2.6, we can switch from a *right* factorization of a spectrum ( $\Phi = H^*H$ ) to a *left* factorization ( $\Phi = WW^*$ ), and vice versa. We will now show that both (4.7) and (4.8) can be reduced to integrals of the form

$$\int G(z)\Delta_{\Lambda}^{-1}(z)\Phi(z)\Delta_{\Lambda}^{-*}(z)G^{*}(z)$$
(4.24)

where  $\Phi(z)$  is a spectrum. Indeed, let  $\Phi_{\Psi}(z) = \Delta_{\Lambda}^{-*}(z)\Psi(z)\Delta_{\Lambda}^{-1}(z)$ . Then

$$\int G(z)Q_{\Lambda}^{-1}(z)\Psi(z)Q_{\Lambda}^{-1}(z)G^{*}(z)$$

$$= \int G(z)\Delta_{\Lambda}^{-1}(z)\left(\Delta_{\Lambda}^{-*}(z)\Psi(z)\Delta_{\Lambda}^{-1}(z)\right)\Delta_{\Lambda}^{-*}(z)G^{*}(z)$$
(4.25)

which has the form (4.24) with  $\Phi = \Phi_{\Psi}$ . Applying Lemma 4.2.5 we obtain a (left) spectral factor of  $\Phi_{\Psi}(z)$ :

$$\Phi_{\Psi}(z) = \Delta_{\Lambda}^{-*}(z)(W_{\Psi}(z)W_{\Psi}^{*}(z))\Delta_{\Lambda}^{-1}(z) 
= \Delta_{\Lambda}^{-*}(z)(H_{\Psi}^{*}(z)H_{\Psi}(z))\Delta_{\Lambda}^{-1}(z) 
= (H_{\Psi}(z)\Delta_{\Lambda}^{-1}(z))^{*}(H_{\Psi}(z)\Delta_{\Lambda}^{-1}(z)) 
= W_{1}(z)W_{1}^{*}(z)$$
(4.26)

Finally, (4.25) can be computed obtaining a realization of  $G(z)\Delta_{\Lambda}^{-1}(z)W_1(z)$  and applying Lemma 4.2.1. Now, let  $\Phi_{\Sigma}(z) = \Delta_{\Lambda}^{-*}(z)G^*(z)\Sigma G(z)\Delta_{\Lambda}^{-1}(z)$ , where  $\Sigma$  is

one of the precomputed generators of  $\operatorname{Range}\Gamma.$  Then

$$\int GQ_{\Lambda}^{-1} \left[ (G^*\Sigma GQ_{\Lambda}^{-1}\Psi) + (G^*\Sigma GQ_{\Lambda}^{-1}\Psi)^* \right] Q_{\Lambda}^{-1}G^*$$

$$= \int G\Delta_{\Lambda}^{-1}\Delta_{\Lambda}^{-*} \left[ (G^*\Sigma G\Delta_{\Lambda}^{-1}\Delta_{\Lambda}^{-*}\Psi) + (\Psi\Delta_{\Lambda}^{-1}\Delta_{\Lambda}^{-*}G^*\Sigma G) \right] \Delta_{\Lambda}^{-1}\Delta_{\Lambda}^{-*}G^*$$

$$= \int G\Delta_{\Lambda}^{-1} \left[ (\Delta_{\Lambda}^{-*}G^*\Sigma G\Delta_{\Lambda}^{-1}\Delta_{\Lambda}^{-*}\Psi\Delta_{\Lambda}^{-1}) + (\Delta_{\Lambda}^{-*}\Psi\Delta_{\Lambda}^{-1}\Delta_{\Lambda}^{-*}G^*\Sigma G\Delta_{\Lambda}^{-1}) \right] \Delta_{\Lambda}^{-*}G^*$$

$$= \int G\Delta_{\Lambda}^{-1} \left[ \Phi_{\Sigma}\Phi_{\Psi} + \Phi_{\Psi}\Phi_{\Sigma} \right] \Delta_{\Lambda}^{-*}G^*$$

$$= \int G\Delta_{\Lambda}^{-1} \left[ (\Phi_{\Sigma} + \Phi_{\Psi})(\Phi_{\Sigma} + \Phi_{\Psi}) - \Phi_{\Psi}\Phi_{\Psi} - \Phi_{\Sigma}\Phi_{\Sigma} \right] \Delta_{\Lambda}^{-*}G^*$$

$$= \int G\Delta_{\Lambda}^{-1} \left[ (\Phi_{\Sigma} + \Phi_{\Psi})(\Phi_{\Sigma} + \Phi_{\Psi}) \right] \Delta_{\Lambda}^{-*}G^*$$

$$= \int G\Delta_{\Lambda}^{-1} \left[ (\Phi_{\Sigma} + \Phi_{\Psi})(\Phi_{\Sigma} + \Phi_{\Psi}) \right] \Delta_{\Lambda}^{-*}G^*$$

which is a difference of integrals of the form (4.24). To compute (4.27), we must obtain (left) spectral factors of  $\Phi_{\Psi}\Phi_{\Psi}^*$ ,  $\Phi_{\Sigma}\Phi_{\Sigma}^*$  and  $(\Phi_{\Sigma} + \Phi_{\Psi})(\Phi_{\Sigma} + \Phi_{\Psi})^*$ . Suppose, first, that  $\Sigma > 0$ . For the first spectrum we have

$$\Phi_{\Psi} \Phi_{\Psi}^* = W_1(W_1^* W_1) W_1^* = W_1(H_1 H_1^*) W_1^*$$
  
=  $(W_1 H_1) (W_1 H_1)^*$  (4.28)

For the second, we have

$$\Phi_{\Sigma} = (\Delta_{\Lambda}^{-*} G^* \Sigma^{1/2}) (\Sigma^{1/2} G \Delta_{\Lambda}^{-1}) = H_{\Sigma}^* H_{\Sigma} = W_{\Sigma} W_{\Sigma}^*$$
  

$$\Phi_{\Sigma} \Phi_{\Sigma}^* = W_{\Sigma} (W_{\Sigma}^* W_{\Sigma}) W_{\Sigma}^* = W_{\Sigma} (K_{\Sigma} K_{\Sigma}^*) W_{\Sigma}^*$$
  

$$= (W_{\Sigma} K_{\Sigma}) (W_{\Sigma} K_{\Sigma})^*$$
(4.29)

And for the third:

$$\begin{aligned} (\Phi_{\Sigma} + \Phi_{\Psi})(\Phi_{\Sigma} + \Phi_{\Psi})^{*} \\ &= (Z_{\Sigma} + Z_{\Sigma}^{*} + Z_{1} + Z_{1}^{*})(Z_{\Sigma} + Z_{\Sigma}^{*} + Z_{1} + Z_{1}^{*})^{*} \\ &= ((Z_{\Sigma} + Z_{1}) + (Z_{\Sigma} + Z_{1})^{*})((Z_{\Sigma} + Z_{1}) + (Z_{\Sigma} + Z_{1})^{*})^{*} \\ &= (Z_{1\Sigma} + Z_{1\Sigma}^{*})(Z_{1\Sigma} + Z_{1\Sigma}^{*})^{*} \\ &= W_{1\Sigma}(W_{1\Sigma}^{*}W_{1\Sigma})W_{1\Sigma}^{*} \\ &= W_{1\Sigma}(H_{1\Sigma}H_{1\Sigma}^{*})W_{1\Sigma}^{*} \\ &= (W_{1\Sigma}H_{1\Sigma})(W_{1\Sigma}H_{1\Sigma})^{*} \end{aligned}$$
(4.30)

where  $Z_1$  is the causal part of  $\Phi_{\Psi}$ ,  $Z_{1\Sigma} = Z_1 + Z_{\Sigma}$ ,  $W_{1\Sigma}$  is a left factor of the spectrum  $Z_{1\Sigma} + Z_{1\Sigma}^*$ ,  $H_{1\Sigma}$  is a left factor of the spectrum  $W_{1\Sigma}^*W_{1\Sigma}$ , and where we used Lemma 4.2.1 to obtain the causal part of  $\Phi_{\Sigma}$  and  $\Phi_{\Psi}$  from their spectral factors, and Lemma 4.2.2 to obtain the minimum phase spectral factor of the sum  $\Phi_{\Sigma} + \Phi_{\Psi}$  from its causal part. Thus, if  $\Sigma > 0$ , we really have all the tools to compute integral (4.27).

Now,  $\Sigma$  is *not* necessarily positive definite, but if  $-\lambda < 0$  is the minimum between the eigenvalues of all the generators  $\Sigma_k$ , then  $\Sigma + (\lambda + 1)I$  is positive definite. Thus, in the general case, by linearity (4.27) can be reduced to:

$$\int G\Delta_{\Lambda}^{-1} \left[ \Phi_{\Sigma} \Phi_{\Psi} + \Phi_{\Psi} \Phi_{\Sigma} \right] \Delta_{\Lambda}^{-*} G^{*}$$

$$= \int G\Delta_{\Lambda}^{-1} \left[ \Phi_{\Sigma+(\lambda+1)I-(\lambda+1)I} \Phi_{\Psi} + \Phi_{\Psi} \Phi_{\Sigma+(\lambda+1)I} \right] \Delta_{\Lambda}^{-*} G^{*} \qquad (4.31)$$

$$= \int G\Delta_{\Lambda}^{-1} \left[ \Phi_{\Sigma+(\lambda+1)I} \Phi_{\Psi} + \Phi_{\Psi} \Phi_{\Sigma+(\lambda+1)I} \right] \Delta_{\Lambda}^{-*} G^{*} - (\lambda+1) \int G\Delta_{\Lambda}^{-1} \left[ \Phi_{I} \Phi_{\Psi} + \Phi_{\Psi} \Phi_{I} \right] \Delta_{\Lambda}^{-*} G^{*}$$

which is a difference of integrals with the same structure of (4.27), and that are computable with the above tools (obviously  $\int G\Delta_{\Lambda}^{-1} [\Phi_I \ \Phi_{\Psi} + \Phi_{\Psi} \ \Phi_I] \Delta_{\Lambda}^{-*} G^*$  needs to be computed only once). This enables us to solve equation (4.5).

### 4.2.3 Computations in the backtracking step

The backtracking stage involves similar, though easier, computations. We must check the following conditions:

$$\Lambda_i + t_i^k \Delta \Lambda_i \in \mathcal{L}_{\Gamma}^H \tag{4.32}$$

$$J_{\Psi}(\Lambda_i + t_i^k \Delta \Lambda_i) < J_{\Psi}(\Lambda_i) + \alpha t_i^k \nabla J_{\Psi \Lambda_i} \Delta \Lambda_i$$
(4.33)

Checking (4.32) is really a matter of checking whether we can factorize  $I + G^*(\Lambda_i + t_i^k \Delta \Lambda_i)G$ . Thus  $t_i^k$  must be halved until the ARE (4.21) is solvable having  $\Lambda = \Lambda_i + t_i^k \Delta \Lambda_i$ .

Finally, to check (4.33), we need to compute  $J_{\Psi}$ . This can be done in a way similar

to the above computations:

$$J_{\Psi}(\Lambda) = \operatorname{tr} \int (I + G^* \Lambda G)^{-1} \Psi + \operatorname{tr} \Lambda$$
  

$$= \operatorname{tr} \int \Delta_{\Lambda}^{-1} \Delta_{\Lambda}^{-*} W_{\Psi} W_{\Psi}^* + \operatorname{tr} \Lambda$$
  

$$= \operatorname{tr} \int \Delta_{\Lambda}^{-*} (W_{\Psi} W_{\Psi}^*) \Delta_{\Lambda}^{-1} + \operatorname{tr} \Lambda$$
  

$$= \operatorname{tr} \int \Delta_{\Lambda}^{-*} (H_{\Psi}^* H_{\Psi}) \Delta_{\Lambda}^{-1} + \operatorname{tr} \Lambda$$
  

$$= \operatorname{tr} \int (H_{\Psi} \Delta_{\Lambda}^{-1})^* (H_{\Psi} \Delta_{\Lambda}^{-1}) + \operatorname{tr} \Lambda$$
  

$$= \operatorname{tr} \int W W^* + \operatorname{tr} \Lambda$$
  
(4.34)

## 4.3 **Proof of global convergence**

Given that the minimum of  $J_{\Psi}$  exists and is unique, we investigate global convergence of our Newton algorithm. First, we recall the following

Definition: a function f(x) twice differentiable in a set S is said to be strongly convex in S if there exists a constant m > 0 such that  $H(x) \ge mI$  for  $x \in S$ , where H(x) is the Hessian of f at x.

We restrict our analysis to a sublevel set of  $J_{\Psi}$ . Let  $\Lambda_0 = 0$ . The set

$$S := \left\{ \Lambda \in \mathcal{L}_{\Gamma}^{H} : J_{\Psi}(\Lambda) \le J_{\Psi}(\Lambda_{0}) = \operatorname{tr} \int \Psi \right\}$$
(4.35)

is compact (as it was shown in [21, Section VII]). Because of the backtracking in the algorithm, the sequence  $J_{\Psi}(\Lambda_0), J_{\Psi}(\Lambda_1), \dots$  is decreasing. Thus  $\Lambda_n \in S, \forall n \geq 0$ . We now wish to apply a theorem in [4, 9.5.3, p. 488] on convergence of the Newton algorithm with backtraking for strongly convex functions on  $\mathbb{R}^n$ . This theorem ensures linear decrease for a finite number of steps, and quadratic convergence to the minimum after the linear stage, thus establishing *global* convergence of the Newton algorithm with backtracking. We proceed to establish first strong convexity of  $J_{\Psi}$  on S. To do that, we employ the following result.

**Lemma 4.3.1** Let f(x) be defined over an open convex subset D of a finite-dimensional linear space V. Assume that f is twice continuously differentiable and strictly convex on D. Then f is strongly convex on any compact set  $S \subset D$ .

#### Proof.

First, recall that since f is twice continuously differentiable and strictly convex, its Hessian  $H_x$  is an Hermitian positive-definite matrix at each point x. By Lemma 3.5.6, the mapping from H to its minimum (real) eigenvalue is continuous. It follows that the mapping from x to the minimum eigenvalue of the Hessian of f at x is also continuous, being a composition of continuous functions. Hence the latter admits a minimum m in the compact set S by Weierstrass' theorem. Thus m is the minimum of the eigenvalues of all the Hessians computed in S, and m cannot be zero, since otherwise there would be an x with  $H_x$  singular, and this cannot happen since f is strictly convex. Hence  $H_x \ge mI \ \forall x \in S$ , i.e. f is strongly convex on S.

**Remark 4.3.2** By an argument similar to that of Lemma 3.5.6, it can be shown that for a twice continuously differentiable function which is strictly convex on D, there exists M > 0 such that  $H_x \leq MI$  for all  $x \in S$ . Moreover, strong convexity on a closed set S implies boundedness of the latter. Thus, strong convexity and boundedness of the Hessian are intertwined, and both are essential in the proof of Theorem 4.3.3 (see [4]).

**Theorem 4.3.3** The following facts hold true:

- 1.  $J_{\Psi}$  is twice continuously differentiable on S;
- 2.  $J_{\Psi}$  is strongly convex on S;
- 3. the Hessian of  $J_{\Psi}$  is Lipschitz-continuous over S;
- 4. the sequence  $\{\Lambda_i; i \geq 0\}$  generated by the Newton algorithm of Section 4.2 (4.4)-(4.33) converges to the unique minimum point of  $J_{\Psi}$  in  $\mathcal{L}_{\Gamma}^H$ .

#### Proof.

Property 1 is a trivial consequence of Theorem 3.5.8. To prove 2, remember that  $J_{\Psi}$  is strictly convex on  $\mathcal{L}_{\Gamma}^{H}$ , hence also on S, and apply Lemma 4.3.1. As for property 3, what it really says is that the following operator:

$$\mathcal{H}: \Lambda \mapsto H_{\Lambda}(\cdot, \cdot)$$

is Lipschitz continuous on S. Theorem 3.5.8 implies that  $J_{\Psi} \in C^3(\mathcal{L}_{\Gamma}^H)$  or, which is the same, that  $\mathcal{H} \in C^1(\mathcal{L}_{\Gamma}^H)$ . The continuous differentiability of  $\mathcal{H}$  implies its Lipschitz continuity over an arbitrary compact subset of  $\mathcal{L}_{\Gamma}^H$ , hence also over the sublevel set S, and property 3 follows. Finally, to prove 4, notice that all the hypotheses of [4, 9.5.3, p. 488] are satisfied. Namely, the function to be minimized  $J_{\Psi}$  is strongly convex on the compact set S, and its Hessian is Lipschitz-continuous over S. It remains to observe that  $J_{\Psi}$  is defined over a subset of the linear space Range  $\Gamma$  which has *finite* dimension d over  $\mathbb{R}$  (recall that Range  $\Gamma$  is spanned by a finite set of matrices. See Proposition 1.3.1 and Remark 1.3.3, where  $d \leq N$ ). Thus, once we choose a base in Range  $\Gamma$ , to every  $\Lambda \in \mathcal{L}_{\Gamma}^H$  there corresponds a vector in  $\mathbb{R}^d$ , to every positive definite bilinear form over Range  $\Gamma$  there corresponds a compact set in  $\mathbb{R}^d$ . Hence, every convergence result that holds in  $\mathbb{R}^d$  must also hold in the abstract setting, in view of the homeomorphism between one space and the other.

# Chapter 5

# An application to spectrum estimation

## 5.1 A spectral estimation procedure

Following the purposes of the THREE method presented in [7], now we describe an application of the above approximation algorithm to the estimation of spectral densities. Consider first the scalar case, and suppose that the finite sequence  $y_1, ..., y_N$ is extracted from a realization of a zero-mean, weakly stationary discrete-time process  $\{y_t\}_{t=-\infty}^{+\infty}$ . We want to estimate the spectral density  $\Phi_y(e^{j\vartheta})$  of y. The idea is the following:

- Fix a transfer function  $G(z) = (zI A)^{-1}B$ , feed the data  $\{y_i\}$  to it, and collect the output data  $\{x_i\}$ .
- Compute a consistent, and possibly unbiased, estimate  $\hat{\Sigma}$  of the covariance matrix of the outputs  $\{x_i\}$ . Note that some output samples  $x_1, ..., x_M$  should be discarded so that the filter can be considered to operate in steady state.
- Choose as "prior" spectrum  $\hat{\Phi}_y$  a coarse, low-order, estimate of the true spectrum of y obtained by means of another (simple) identification method.
- "Refine" the estimate  $\hat{\Phi}_y$  by solving the approximation problem (3.5.1) with respect to G(z),  $\Sigma = \hat{\Sigma}$ , and  $\Psi = \hat{\Phi}_y$  ("rescaling" G(z) in order to obtain  $\Sigma \to I$ , as in Remark 1.2.1).

To be clear, the result of the above procedure is the only spectrum, compatible with the output variance  $\hat{\Sigma}$ , which is closest to the rough estimate  $\hat{\Phi}_y$  in the  $d_H$  distance. Note that we are left with significant degrees of freedom in applying the above procedure: The method for estimating  $\hat{\Phi}_y$ , in particular its degree, and the whole structure of  $G(z) = (zI - A)^{-1}B$ , which has no contraints other than A being a stability matrix and (A, B) being reachable. The coarsest possible estimate of  $\Phi_y$  is the constant spectrum equal to the sample variance of the  $\{y_i\}$ , i.e.  $\hat{\Phi}_y(e^{j\vartheta}) \equiv \hat{\sigma}_y^2$ , where  $\hat{\sigma}_y^2 = \frac{1}{N-1} \sum_{i=1}^N |y_i|^2$ . The resulting spectrum has the form  $\hat{\sigma}_y^2(1 + G^*\hat{\Lambda}G)^{-2}$ . Another simple choice is  $\hat{\Phi}_y = W(z)W^*(z)$ , where  $W(z) = \hat{\sigma}_e \frac{c(z)}{a(z)}$  is a low-order AR, MA or ARMA model estimated from  $y_1, ..., y_N$  by means of predictive error minimization methods or the like.

The flexibility in the choice of G(z) is more essential, and has more profound implications. As described in [7], [34], [22] and [21], Moreover, the following choice:

$$A = \begin{bmatrix} 0 & 1 & 0 & \dots & 0 \\ 0 & 0 & 1 & \dots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \dots & 1 \\ 0 & 0 & 0 & \dots & 0 \end{bmatrix}, \quad B = \begin{bmatrix} 0 \\ 0 \\ \vdots \\ 0 \\ 1 \end{bmatrix}$$
(5.1)

implies that the steady-state variance  $\Sigma$  is a Toeplitz matrix whose diagonals contain the lags  $c_0, c_1, ..., c_{n-1}$  of the covariance signal of the input, and the corresponding problem of finding *any* spectrum that satisfies (3.52) is a covariance extension problem.

These facts justify the theoretical interest in algorithms for constrained spectrum approximation, if for no other reason, as tools to compute at least *one* solution to a Nevanlinna-Pick interpolation or to a covariance extension problem, respectively. But the freedom in choosing G(z) has implications also in the above practical application to spectral estimation, where the key properties, not surprisingly, depend on the poles of G(z), i.e., the eigenvalues of A. In general, as described in [7], the *magnitude* of the latter has implications on the variance of the sample covariance  $\hat{\Sigma}$ : The closer the eigenvalues to the origin, the smaller that variance (see [7, Section II.D]). Moreover, at least as far as THREE [7] is concerned, the *phase* of the eigenvalues influences resolution capability: More precisely, the spectrum estimation procedure has higher resolution in those sectors of the unit circle where more eigenvalues are located. According to simulations, the latter statement appears to be true also in our setting (the fundamental difference being that the metric which is minimized is the Hellinger distance instead of the Kullback-Leibler one).

**Remark 5.1.1** In the above setting  $\hat{\Sigma}$  is a consistent estimate of the true steady-state variance. Although  $\hat{\Sigma}$  must belong to Range  $\Gamma$  as  $N \to +\infty$  (this being the case even if y is the sum of a purely nondeterministic process and some sinusoids, as in the simulations that follow), it is almost certainly not the case that  $\hat{\Sigma} \in \text{Range }\Gamma$  when we have available only the finitely many data  $x_{M+1}, ..., x_N$ . Strictly speaking, this implies that the contraint (3.52) with  $\Sigma = \hat{\Sigma}$  is almost always not feasible. It turns out that, increasing the tolerance threshold in its step 5, the Newton algorithm exhibits some kind of robustness in this respect. That is, it leads to a  $\Lambda$  whose corresponding spectrum  $\hat{\Phi}$  is close to satisfying the constraint.

Nevertheless, we prefer a clear understanding of what the resulting spectrum really is. Thus, we choose to enforce feasibility of the approximation problem, at least as permitted by machine number representation, before starting the optimization procedure.

#### 5.2. SIMULATION RESULTS

To this end, following the same approach employed in [7], we pose the approximation problem not in terms of the estimated  $\hat{\Sigma}$ , but in terms of its orthogonal projection  $\hat{\Sigma}_{\Gamma}$ onto Range  $\Gamma$ , which can be easily computed by means of algebraic methods. That is to say: We cannot approximate in the preimage  $\Gamma^{-1}(\hat{\Sigma})$ , because that set is empty, thus we choose to approximate in  $\Gamma^{-1}(\hat{\Sigma}_{\Gamma})$ , where  $\hat{\Sigma}_{\Gamma}$  is the matrix closest to  $\hat{\Sigma}$  such that its preimage is not empty. This seems a reasonable choice and by the way it is, mutatis mutandis, what the Moore-Penrose pseudoinverse does for the "solution"  $\hat{x} = A^{\dagger}b$ , when the linear system Ax = b is not solvable.

Note that it is not guaranteed at all that the projection of a positive definite matrix onto a subspace of the Hermitian matrices is itself positive definite. In practice, this is not really a problem, inasmuch  $\hat{\Sigma}$  is "sufficiently positive" and close to Range  $\Gamma$ . The positivity of  $\hat{\Sigma}_{\Gamma}$  must anyway be checked before proceeding. This approach and the considerations on the positivity issue should be compared to [7, Section II.D], which deals with the particular case when Range  $\Gamma$  is the space of Toeplitz matrices, and to [31, Section 4], where, to find a matrix a  $\hat{\Sigma}_{\Gamma}$  close to  $\hat{\Sigma}$ , a Kullback-Leibler criterion is adopted instead of least squares.

## 5.2 Simulation results

#### 5.2.1 Simulation results: Scalar case



Figure 5.1: Estimation of an ARMA(6,4) spectrum by means of Hellinger-distance spectrum approximation, constant prior and AR(3) prior.

Figure 5.1 shows the results of the above estimation procedure with G(z) structured according to the covariance extension setting (5.1) with 6 covariance lags (i.e.

n = 6, A is  $6 \times 6$ ), run over 500 samples of the following ARMA process:

$$y(t) = 0.5y(t-1) - 0.42y(t-2) + 0.602y(t-3) - 0.0425y(t-4) + 0.1192y(t-5) + e(t) + 1.1e(t-1) + 0.08e(t-2) - 0.15e(t-3)$$

(poles in 0.9,  $-0.2 \pm 0.7$ j,  $\pm 0.5$ j) where e(t) is a zero-mean Gaussian white noise with unit variance. Two priors, both estimated from data, have been considered: the constant spectrum  $\hat{\Phi}_y(e^{j\vartheta}) \equiv \hat{\sigma}_y^2$  and the spectrum  $\hat{\Phi}_y = W_{AR}(z)W_{AR}^*(z)$ , where  $W_{AR}(z) = \frac{\hat{\sigma}_e}{a(z)}$  is an AR model of order 3 obtained from the data by means of the Predictive Error Method procedure in Matlab's System Identification toolbox.



Figure 5.2: Spectral estimates of two sinusoids with superimposed noise by means of Hellinger-distance spectrum approximation, constant prior. Compare with [7, Section IV.B, Example 1].

Figure 5.2 shows the performance of the above procedure in a setting that resembles that of [7, Section IV.B, Example 1]. The estimation procedure was run on 300 samples of a superposition of two sinusoids in colored noise:

$$y(t) = 0.5\sin(\omega_1 t + \phi_1) + 0.5\sin(\omega_2 t + \phi_2) + z(t)$$
  
$$z(t) = 0.8z(t-1) + 0.5\nu(t) + 0.25\nu(t-1)$$

with  $\phi_1$ ,  $\phi_2$  and  $\nu(t)$  independent normal random variables with zero mean and unit variance,  $\omega_1 = 0.42$  and  $\omega_2 = 0.53$ . The prior here considered is the constant spectrum equal to the sample variance of the  $\{y_i\}$  data. Following [7], A was chosen real blockdiagonal with the following poles (equispaced in a narrow range where the frequencies of the two sinusoids lie, to increase resolution in that region):

 $\begin{array}{l} 0,0.85,-0.85,\\ 0.9\mathrm{e}^{\pm\mathrm{j}0.42},0.9\mathrm{e}^{\pm\mathrm{j}0.44},0.9\mathrm{e}^{\pm\mathrm{j}0.46},0.9\mathrm{e}^{\pm\mathrm{j}0.48},0.9\mathrm{e}^{\pm\mathrm{j}0.50} \end{array}$ 

(and *B* a column of ones). It can be seen that Hellinger-distance based approximation does a good job, as does the THREE algorithm, at detecting the spectral lines at frequencies  $\omega_1$  and  $\omega_2$ .

#### 5.2.2 Simulation results: Multivariate case

We now consider spectral estimation for a multivariate process. Here, 100 samples of a bivariate process with a high order spectrum were generated by feeding a bivariate Gaussian white noise with mean 0 and variance I to a square (stable) shaping filter of order 40. The latter was constructed with random coefficients, except for one fixed conjugate pair of poles with radius 0.9 and argument 0.52, and one fixed conjugate pair of zeros with radius  $1-10^{-5}$  and argument 0.2. The transfer function G(z) was chosen with one pole in the origin and 4 complex pole pairs with radius 0.9 and frequencies equispaced in the range  $[0,\pi]$ . Then the above estimating procedure was applied, with prior spectrum chosen as the constant density equal to the sample covariance of the bivariate process y. Figure 5.3 shows a plot of  $\Phi_{11}(e^{j\vartheta})$ ,  $\operatorname{Re} \Phi_{12}(e^{j\vartheta})$ ,  $\operatorname{Im} \Phi_{12}(e^{j\vartheta})$ and  $\Phi_{22}(e^{j\vartheta})$ , respectively for the true spectrum and for the estimation of the latter based on one run of 100 samples. In Figure 5.4 we compare the performances of various spectral estimation methods in the following way. We consider four estimates  $\hat{\Phi}_{\rm H}, \, \hat{\Phi}_{\rm ME}, \, \hat{\Phi}_{\rm PEM}, \, \text{and} \, \hat{\Phi}_{\rm N4SID} \, \text{of} \, \Phi.$  The spectral density  $\hat{\Phi}_{\rm H}$  is the estimate obtained by the procedure described above in Subsection 5.1. The spectral density  $\hat{\Phi}_{ME}$  is the maximum entropy estimate [27] obtained using the same G(z) employed to obtain our estimate. The spectral densities  $\Phi_{\text{PEM}}$  and  $\Phi_{\text{N4SID}}$  are the estimates of  $\Phi$  obtained by using "off-the-shelf" Matlab procedures for the Prediction Error Method (see i.e. [55] or [43]) and for the N4SID method (see [58] or [43]): The former is a multivariable extension of the classical approach to ARMAX identification, while the latter is a standard algorithm in the modern field of subspace identification. In order to obtain a comparison reasonably independent of the specific data set, we have performed 50 independent runs each with 100 samples of y. In such a way we have obtained 50 different estimates  $\Phi_{M,i}$ , M = H, ME, PEM, N4SID, i = 1, 2, ..., 50, for each method.

We have then defined

J

$$E_{\rm H}(\vartheta) := \frac{1}{50} \sum_{i=1}^{50} \|\hat{\Phi}_{{\rm H},i}({\rm e}^{j\vartheta}) - \Phi({\rm e}^{j\vartheta})\|, \qquad (5.2)$$

where  $\|\cdot\|$  denotes the spectral norm. This is understood as the average estimation error of our method at each frequency. Similarly, we have defined the average errors  $E_{\text{ME}}(\vartheta)$ ,  $E_{\text{PEM}}(\vartheta)$ , and  $E_{\text{N4SID}}(\vartheta)$  of the other methods. In the each of the plots



Figure 5.3: Estimation of the spectrum of a bivariate process with rich dynamics by means of Hellinger-distance spectrum approximation, constant prior.

of Figure 5.4, we depict the average error of our method  $E_{\rm H}(\vartheta)$  together with the average error of one of the other methods. More explicitly, the first diagram shows the error for the Hellinger approximation method and for the maximum entropy spectrum described in [27]. The second diagram shows the error for the Hellinger approximation and for the spectrum obtained via MATLAB's PEM identification method. The third diagram shows the same for Hellinger approximation and MATLAB's N4SID method. The Hellinger approximation based approach appears to perform better or much better than the other methods. The simulation yields similar results with N = 200 data points. With N = 300 data samples, PEM and N4SID perform as well as our method.

Of course, one should always take into account the complexity of the resulting spectrum. In this example, G(z) being of order 9, the resulting spectral factor (or "model") produced by the Hellinger approximation has order 18, whereas the corresponding maximum entropy model has order 9 and both N4SID and PEM usually choose order 10.

In our simulation, the norm of the difference of two estimates produced by PEM or by N4SID is sometimes very large when compared to the norm of the difference between any two of the estimates produced by our method. That is, although PEM and N4SID are provably consistent as  $N \to \infty$ , when few data are available both of them may introduce occasional artifacts, which are well visible as "peaks" in figure



Figure 5.4: Estimation of the spectrum of a bivariate process with rich dynamics by means of various methods. Comparison between the spectral norm of the differences  $\hat{\Phi}_{\rm H} - \Phi$ ,  $\hat{\Phi}_{\rm ME} - \Phi$ ,  $\hat{\Phi}_{\rm PEM} - \Phi$ , and  $\hat{\Phi}_{\rm N4SID} - \Phi$  (average over 50 simulations).

5.4 (a "peak" in the 50-run average is due to a very high error in one of the runs, not to a systematic error). Our method appears to be more robust in this respect.

# Appendix A

# Proofs of some technical lemmas

**Lemma A.0.1** Let  $H \in \mathcal{H}(n)$  and m be its minimum eigenvalue. The map  $H \mapsto m$  is continuous.

*Proof.* The map from a matrix H to the vector of coefficients of its characteristic polynomial  $a(s) = \det(sI - H) = a_0 + \ldots + a_{n-1}s^{n-1} + s^n$  is continuous. Indeed, each of the coefficients of a(s) is obtained by means of sums and products of elements of H. Moreover, it is a well-known fact (see for example [45]) that the mapping from the coefficients of a monic polynomial to its roots is continuous, in the following sense: Given  $a(s) = s^n + \sum_{i=0}^{n-1} a_i s^i$ , let  $\lambda_i$  be the zeros of a(s) and  $\nu_i$  the respective multiplicities. For all  $\varepsilon > 0$ , there exists  $\delta > 0$  such that if  $b(s) = s^n + \sum_{i=0}^{n-1} b_i s^i$  and  $|b_i - a_i| < \delta$  for all  $i = 0, 1, \ldots, n-1$ , then b(s) has  $\nu_i$  zeros in the ball centered in  $\lambda_i$  with radius  $\varepsilon$ . In conclusion, if H is Hermitian, the mapping from H to its minimum (real) eigenvalue is continuous.

**Lemma A.0.2** Let A be a stability matrix and  $H(z) = C(zI - A)^{-1}B + D$  be a minimal realization. Let P be the solution of the Lyapunov equation

$$P - A^* P A = C^* C. \tag{A.1}$$

Let  $\begin{bmatrix} K \\ J \end{bmatrix}$  be an ortho-normal basis of ker  $\begin{bmatrix} A^*P^{1/2} & C^* \end{bmatrix}$  i.e.

$$\begin{bmatrix} A^* P^{1/2} & C^* \end{bmatrix} \begin{bmatrix} K \\ J \end{bmatrix} = 0, \qquad \begin{bmatrix} K^* & J^* \end{bmatrix} i \begin{bmatrix} K \\ J \end{bmatrix} = I.$$
(A.2)

Let  $G := P^{-1/2}K$  and

$$H_1(z) := (D^*C + B^*PA)(zI - A)^{-1}G + B^*PG + D^*J.$$
(A.3)

Then,  $H^*H = H_1H_1^*$ .

*Proof.* Let  $Q := C(zI - A)^{-1}G + J$ . We first show that  $QQ^* = I$ , so that Q is inner. We then prove that  $Q^*H = H_1^*$ , concluding the proof. We have

$$Q^*Q = G^*(z^{-1}I - A^*)^{-1}C^*C(zI - A)^{-1}G + G^*(z^{-1}I - A^*)^{-1}C^*J + J^*C(zI - A)^{-1}G + J^*J$$
(A.4)

Now let P > 0 be the solution of the Lyapunov equation (A.1). Then,

$$C^*C = -(z^{-1}I - A^*)P(zI - A) + + (z^{-1}I - A^*)Pz + z^{-1}P(zI - A)$$
(A.5)

Substituting (A.5) into (A.4) we obtain

$$Q^{*}Q = -G^{*}PG + G^{*}Pz(zI - A)^{-1}G + G^{*}(z^{-1}I - A^{*})^{-1}z^{-1}PG + G^{*}(z^{-1}I - A^{*})^{-1}C^{*}J + J^{*}C(zI - A)^{-1}G + J^{*}J$$
(A.6)

Moreover,

$$z(zI - A)^{-1} = I + A(zI - A)^{-1} \text{ and} (z^{-1}I - A^*)^{-1}z^{-1} = I + (z^{-1}I - A^*)^{-1}A^*$$
(A.7)

so that

$$Q^*Q = (J^*C + G^*PA)(zI - A)^{-1}G + + ((J^*C + G^*PA)(zI - A)^{-1}G)^* + G^*PG + J^*J$$
(A.8)

Taking (A.2) into account, it is easy to see that  $Q^*Q = I$ . Therefore,  $H^*H =$ 

 $H^*QQ^*H$ . Recalling (A.5) and (A.7), we eventually get

$$\begin{split} Q^*H &= \\ &= (G^*(z^{-1}I - A^*)^{-1}C^* + J^*)(C(zI - A)^{-1}B + D) \\ &= -G^*(z^{-1}I - A^*)^{-1}(z^{-1}I - A^*)P(zI - A)(zI - A)^{-1}B \\ &+ G^*(z^{-1}I - A^*)^{-1}z^{-1}P(zI - A)(zI - A)^{-1}B \\ &+ G^*(z^{-1}I - A^*)^{-1}C^*D + J^*C(zI - A)^{-1}B + J^*D \\ &= -G^*PB + G^*Pz(zI - A)^{-1}B \\ &+ G^*(z^{-1}I - A^*)^{-1}z^{-1}PB \\ &+ G^*(z^{-1}I - A^*)^{-1}C^*D + J^*C(zI - A)^{-1}B + J^*D \\ &= -G^*PB + G^*P\left(I + A(zI - A)^{-1}\right)B \\ &+ G^*(z^{-1}I - A^*)^{-1}C^*D + J^*C(zI - A)^{-1}B + J^*D \\ &= G^*PB + G^*PA(zI - A)^{-1}B \\ &+ G^*(z^{-1}I - A^*)^{-1}C^*D + J^*C(zI - A)^{-1}B + J^*D \\ &= G^*PB + G^*PA(zI - A)^{-1}B \\ &+ G^*(z^{-1}I - A^*)^{-1}C^*D + J^*C(zI - A)^{-1}B + J^*D \\ &= G^*(z^{-1}I - A^*)^{-1}C^*D + J^*C(zI - A)^{-1}B + J^*D \\ &= G^*(z^{-1}I - A^*)^{-1}C^*D + J^*C(zI - A)^{-1}B + J^*D \\ &= G^*(z^{-1}I - A^*)^{-1}(C^*D + A^*PB) \\ &+ (G^*PA + J^*C)(zI - A)^{-1}B + G^*PB + J^*D \\ &= H_1^* \end{split}$$

# Bibliography

- S. Amari. Differential-Geometrical Methods in Statistics. Springer-verlag, Heidelberg, 1985.
- [2] A. Barron. Entropy and the central limit theorem. Ann. Probab., 14:336–342, 1986.
- [3] A. Blomqvist, A. Lindquist, and R. Nagamune. Matrix-valued Nevanlinna-Pick interpolation with complexity constraint: An optimization approach. *IEEE Trans. Aut. Control*, 48:2172–2190, 2003.
- [4] S. Boyd and L. Vandenberghe. *Convex Optimization*. Cambridge University Press, Cambridge, UK, 2004.
- [5] C. I. Byrnes, P. Enqvist, and A. Linquist. Cepstral coefficients, covariance lags and pole-zero models for finite data strings. *IEEE Trans. Sig. Proc.*, 50:677–693, 2001.
- [6] C. I. Byrnes, P. Enqvist, and A. Linquist. Identifiability and well-posedness of shaping-filter parameterizations: A global analysis approach. SIAM J. Control and Optimization, 41:23–59, 2002.
- [7] C. I. Byrnes, T. Georgiou, and A. Lindquist. A new approach to spectral estimation: A tunable high-resolution spectral estimator. *IEEE Trans. Sig. Proc.*, 49:3189–3205, 2000.
- [8] C. I. Byrnes, T. Georgiou, and A. Lindquist. A generalized entropy criterion for Nevanlinna-Pick interpolation with degree constraint. *IEEE Trans. Aut. Control*, 46:822–839, 2001.
- C. I. Byrnes, T. Georgiou, A. Lindquist, and A. Megretski. Generalized interpolation in H-infinity with a complexity constraint. *Trans. American Math. Society*, 358(3):965–987, 2006.
- [10] C. I. Byrnes, S. Gusev, and A. Lindquist. A convex optimization approach to the rational covariance extension problem. SIAM J. Control and Opimization, 37:211–229, 1999.

- [11] C. I. Byrnes, S. Gusev, and A. Lindquist. From finite covariance windows to modeling filters: A convex optimization approach. *SIAM Review*, 43:645–675, 2001.
- [12] C. I. Byrnes and A. Lindquist. The generalized moment problem with complexity constraint. *Integral Equations and Operator Theory*, 56(2):163–180, 2006.
- [13] E. Carlen and A. Soffer. Entropy production by convolution and central limit theorems with strong rate information. *Comm. Math. Phys.*, 140:339–371, 1991.
- [14] P. Colaneri and A. Ferrante. Algebraic Riccati equation and J-spectral factorization for  $H_{\infty}$  estimation. Systems & control letters, 51:383–393, 2004.
- [15] T. M. Cover and J. A. Thomas. Information Theory. Wiley, New York, 1991.
- [16] I. Csiszár. Maxent, mathematics and information theory. In K. M. Hanson and R. N. Silver, editors, Proc. 15th Inter. Workshop on Maximum Entropy and Bayesian Methods, pages 35–50, Kos, Greece, 1996. Kluver Academic.
- [17] M. Deistler. A Birds Eye View on System Identification. In A. Chiuso, A. Ferrante, and S. Pinzoni, editors, *Modeling, Estimation and Control: Festschrift in honor of Giorgio Picci on the occasion of his sixty-fifth birthday*, pages 59–71. Springer-Verlag, 2007.
- [18] P. Enqvist. A homotopy approach to rational covariance extension with degree constraint. Int. J. Appl. Math. and Comp. Sci., 11:1173–1201, 2001.
- [19] P. Faurre, M. Clerget, and F. Germain. Operateurs Rationnels Positifs. Dunod, Paris, 1979.
- [20] A. Ferrante, M. Pavon, and F. Ramponi. Further results on the Byrnes-Georgiou-Lindquist generalized moment problem. In A. Chiuso, A. Ferrante, and S. Pinzoni, editors, *Modeling, Estimation and Control: Festschrift in honor of Giorgio Picci on the occasion of his sixty-fifth birthday*, pages 73–83. Springer-Verlag, 2007.
- [21] A. Ferrante, M. Pavon, and F. Ramponi. Hellinger vs. Kullback-Leibler multivariable spectrum approximation. *IEEE Trans. Aut. Control*, 53:954–967, 2008.
- [22] A. Ferrante, M. Pavon, and F. Ramponi. Constrained approximation in the Hellinger distance. In *Proceedings of the European Control Conference 2007* (ECC'07), pages 322–327, Kos, Greece, July 2007.
- [23] H. Föllmer. Random fields and diffusion processes. In P. L. Hennequin, editor, École d'Été de Probabilités de Saint-Flour XV-XVII (Lecture notes in mathematics, vol. 1362), volume 1362, pages 102–203. Springer-Verlag, New York, 1988.

- [24] T. Georgiou. Realization of power spectra from partial covariance sequences. *IEEE Trans. on Acoustics, Speech, and Signal Processing*, 35:438–449, 1987.
- [25] T. Georgiou. The interpolation problem with a degree constraint. *IEEE Trans. Aut. Control*, 44:631–635, 1999.
- [26] T. Georgiou. Spectral estimation by selective harmonic amplification. IEEE Trans. Aut. Control, 46:29–42, 2001.
- [27] T. Georgiou. Spectral analysis based on the state covariance: the maximum entropy spectrum and linear fractional parameterization. *IEEE Trans. Aut. Control*, 47:1811–1823, 2002.
- [28] T. Georgiou. The structure of state covariances and its relation to the power spectrum of the input. *IEEE Trans. Aut. Control*, 47:1056–1066, 2002.
- [29] T. Georgiou. Solution of the general moment problem via a one-parameter imbedding. *IEEE Trans. Aut. Control*, 50:811–826, 2005.
- [30] T. Georgiou. Relative entropy and the multivariable multidimensional moment problem. *IEEE Trans. Inform. Theory*, 52:1052–1066, 2006.
- [31] T. Georgiou. Distances Between Time-Series and Their Autocorrelation Statistics. In A. Chiuso, A. Ferrante, and S. Pinzoni, editors, *Modeling, Estimation and Control: Festschrift in honor of Giorgio Picci on the occasion of his sixty-fifth birthday*, pages 123–133. Springer-Verlag, 2007.
- [32] T. Georgiou. Distances and Riemannian Metrics for Spectral Density Functions. *IEEE Trans. on Signal Processing*, 55(8):3995–4003, August 2007.
- [33] T. Georgiou. An intrinsic metric for power spectral density functions. IEEE Signal Processing Letters, 14(8):561–563, August 2007.
- [34] T. Georgiou and A. Lindquist. Kullback-Leibler approximation of spectral density functions. *IEEE Trans. Inform. Theory*, 49:2910–2917, 2003.
- [35] T. Georgiou and A. Lindquist. Remarks on control design with degree constraint. *IEEE Trans. Aut. Control*, AC-51:1150–1156, 2006.
- [36] T. Georgiou and A. Lindquist. A convex optimization approach to ARMA modeling. *IEEE Trans. Aut. Control*, to appear.
- [37] U. Grenander and G. Szegö. Toeplitz Forms and Their Applications. University of California Press, Berkeley, CA, 1958.
- [38] H. Helson. Lectures on Invariant Subspaces. Academic Press, New York, 1964.

- [39] E. T. Jaynes. Papers on Probability, Statistics and Statistical Physics. R.D. Rosenkranz ed., Dordrecht, 1983.
- [40] P. Kosmol. Optimierung und Approximation. de Gruyter, Berlin, 1991.
- [41] S. Kullback. Information Theory and Statistics 2nd ed. Dover, Mineola NY, 1968.
- [42] A. Lindquist. Prediction-error approximation by convex optimization. In A. Chiuso, A. Ferrante, and S. Pinzoni, editors, *Modeling, Estimation and Con*trol: Festschrift in honor of Giorgio Picci on the occasion of his sixty-fifth birthday, pages 265–275. Springer-Verlag, 2007.
- [43] L. Ljung. System Identification: Theory for the User. Prentice Hall, Englewood Cliffs, 1987.
- [44] S. Luo and Q. Zhang. Informational distance on quantum-state space. Physical Review A, 69, 2004.
- [45] M. Marden. Geometry of polynomials. American Mathematical Society, Providence, RI, 1966.
- [46] R. Nagamune. A robust solver using a continuation method for Nevanlinna-Pick interpolation with degree constraint. *IEEE Trans. Aut. Contr.*, 48:113–117, 2003.
- [47] A. Nasiri Amini, E. Ebbini, and T. Georgiou. Noninvasive estimation of tissue temperature via high-resolution spectral analysis techniques. *IEEE Trans. on Biomedical Engineering*, 52:221–228, 2005.
- [48] M. A. Nielsen and I. L. Chuang. Quantum Computation and Quantum Information. Cambridge Univ. Press, Cambridge, UK, 2000.
- [49] M. Pavon and A. Ferrante. On the Georgiou-Lindquist approach to constrained Kullback-Leibler approximation of spectral densities. *IEEE Trans. Aut. Control*, 51:639–644, 2006.
- [50] M. Pavon and F. Ticozzi. On entropy production for controlled Markovian evolution. J. Math. Phys., 47, 063301, 2006.
- [51] F. Ramponi, A. Ferrante, and M. Pavon. A globally convergent matricial algorithm for multivariate spectral estimation. *IEEE Trans. Aut. Control*, to appear.
- [52] Yu. A. Rozanov. Stationary Random Processes. Holden-Day, San Francisco, 1967.
- [53] W. Rudin. Principles of Mathematical Analysis 3rd ed. McGraw-Hill, New York, 1976.
- [54] J. J. Sakurai. Modern Quantum Mechanics, revised ed. Addison-Wesley, New York, 1994.
- [55] T. Söderström and P. Stoica. System Identification. Prentice Hall, New York, 1989.
- [56] P. Stoica and R. Moses. Introduction to Spectral Analysis. Prentice Hall, New York, 1997.
- [57] A. A. Stoorvogel and J. H. Van Schuppen. System identification with information theoretic criteria. In S. Bittanti and G. Picci, editors, *Identification, Adaptation, Learning: The Science of Learning Models from Data.* Springer, Berlin-Heidelberg, 1996.
- [58] P. Van Overschee and B. DeMoor. N4SID: Subspace algorithms for the identification of combined deterministic-stochastic systems. *Automatica*, 30:75–93, 1994.
- [59] V. Vedral. The role of relative entropy in quantum information theory. Rev. Mod. Phys, 74:197–213, 2002.
- [60] J. Von Neumann. Mathematical Foundations of Quantum Mechanics. Princeton University Press, Princeton, 1955.